



Terahertz Meets AI: the State of the Art

Arshad Farhad D and Jae-Young Pyun *D

Department of Information and Communication Engineering, Chosun University, Gwangju 61452, Republic of Korea; arshad@chosun.ac.kr * Correspondence: jypyun@chosun.ac.kr

Abstract: Terahertz (THz) is a promising technology for future wireless communication networks, particularly for 6G and beyond. The ultra-wide THz band, ranging from 0.1 to 10 THz, can potentially address the limited capacity and scarcity of spectrum in current wireless systems such as 4G-LTE and 5G. Furthermore, it is expected to support advanced wireless applications requiring high data transmission and quality services, i.e., terabit-per-second backhaul systems, ultra-high-definition streaming, virtual/augmented reality, and high-bandwidth wireless communications. In recent years, artificial intelligence (AI) has been used mainly for resource management, spectrum allocation, modulation and bandwidth classification, interference mitigation, beamforming, and medium access control layer protocols to improve THz performance. This survey paper examines the use of AI in state-of-the-art THz communications, discussing the challenges, potentials, and shortcomings. Additionally, this survey discusses the available platforms, including commercial, testbeds, and publicly available simulators for THz communications. Finally, this survey provides future strategies for improving the existing THz simulators and using AI methods, including deep learning, federated learning, and reinforcement learning, to improve THz communications.

Keywords: Terahertz (THz); artificial intelligence (AI); 6G; THz MAC protocols; 6G and beyond; THz simulators

1. Introduction

5G mobile wireless communications have impacted transportation, healthcare, retail, finance, and manufacturing industries by providing high data rates and lower latency, leading to automation in industry and other businesses, and significantly increasing data traffic and intelligent device usage. 5G significantly improved data throughput, latency, network dependability, energy efficiency, and ultra-massive connectivity. Furthermore, the widespread use of smartphones, Internet of Things (IoT) devices, and new multimedia applications have significantly increased mobile data traffic. Therefore, it has revolutionized high-demand data applications such as video conferencing/streaming, virtual reality (VR), telemedicine, and online gaming.

Figure 1 illustrates the continuous evolution of cellular wireless networks, where it can be seen that the evolution is primarily dependent on the increased traffic demands and network capacity. 1G cellular communication technology, mainly analog-based, was introduced in the 1980s, offering only basic voice communication with limited coverage and low capacity. 1G utilized frequency-division multiple access (FDMA) to divide the available spectrum into multiple channels, each of which could be used by a single user. The 2G, announced in the 1990s, offered voice calls and basic data services such as text messaging. In addition, 2G offered time-division multiple access (TDMA) to divide the available spectrum into time slots, improved voice quality, faster data speeds, and more features than 1G, such as a global system for mobile communications (GSM), general packet radio service (GPRS), enhanced data rates for GSM evolution (EDGE). 3G technology was launched in the 2000s, offering high-speed data services such as web browsing. 3G used code-division multiple access (CDMA), improving performance owing to the use of the same



Citation: Farhad, A.; Pyun, J.-Y. Terahertz Meets AI: the State of the Art. *Sensors* **2023**, *23*, 5034. https://doi.org/10.3390/s23115034

Academic Editor: Anastasios Doulamis

Received: 26 April 2023 Revised: 14 May 2023 Accepted: 22 May 2023 Published: 24 May 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). spectrum by multiple users. 4G was announced in the early 2010s, offering high-speed data services, such as video streaming and cloud gaming using orthogonal frequencydivision multiplexing (OFDM). 5G technology was introduced in the 2020s to provide ultra-high data rates, lower latency, and massive device connectivity compared to 4G. In addition, 5G introduced millimeter-wave (*mmWave*), multiple-input and multiple-output (MIMO), beamforming [1], and network slicing, enabling faster downloads, more reliable connections, and support for IoT and smart cities. In addition, the standardization timeline of 5G with key features is illustrated in Table 1 [2–5]. In the next decade, the demands and needs will continue to grow owing to the diverse application requirements, leading to manifold new use cases where the requirements cannot be met by 5G networks, such as holographic teleportation, remote surgery, and unmanned aerial vehicles (UAVs).



Figure 1. The progression of cellular networks, starting with 1G and advancing to 6G, has been accompanied by introducing various applications specific to each generation.

Table 1. 5G standardization timeline [3–5].

Release	Year	Key Features
15 (Phase 1)	2018	Non-standalone 5G new radio (NR), 5G core network, ultra-reliable low-latency communications (URLLC), evolved mobile broadband (eMBB), massive machine type communications (mMTC), beamforming, mmWave, long-term evolution (LTE)-NR dual connectivity, network slicing.
16 (Phase 2)	2020	Standalone 5G NR, enhanced URLLC, industrial IoT, NR sidelink with a focus on vehicle-to- everything (V2X), MIMO, NR-time sensitive networking (TSN) integration, network automation, NR positioning.
17 (Phase 3)	2022	⁵ G NR evolution, enhanced MIMO, enhanced carrier aggregation, 5G NR in unlicensed spectrum, NR-based satellite communications, NR for private networks, AI/ML-based network automation and optimization, multi-access edge computing, NR-based device-to-device communication.
18 (Phase 4)	2024	Will improve physical random access channel coverage, latency, and signaling overhead in mobility scenarios.

As 5G begins to be widely adopted, research institutions worldwide have started to focus on developing 6G wireless communications, which is projected to be rolled out by 2030 [6]. For instance, the University of Oulu launched the 6G Flagship initiative at the beginning of 2018 to perform research activities and other things that could be important for the launch of 6G [7]. The 6G flagship initiative aims to investigate various use cases, including holograms, multi-sensory communications, Terahertz (THz) technology, and pervasive artificial intelligence (AI). Similarly, the US Federal Commission has also

begun exploring the THz spectrum, issuing licenses to the University at Buffalo and the State University of New York to conduct outdoor experiments for developing improved propagation models at 240 GHz and 1.05 THz with effective radiated power (ERP) of 330 W.

1.1. Methodology

We began this survey with a systematic literature review methodology, as described in [8]. First, we started by looking for THz in conjunction with machine learning (ML)/deep learning (DL)/reinforcement learning (RL) in the abstract of each paper found on widely available databases. Then, we discarded out-of-scope papers and added some papers manually with backward reference searching and cross-citation techniques.

1.2. Scope of the Survey

Several surveys and tutorials have presented the THz communication technology and solution designed for 6G wireless communication systems in recent years. Table 2 focuses on and briefly describes the major topics addressed in the existing survey articles and tutorials currently available on THz communication technology. Mainly, these survey articles focus on antenna designs, medium access control (MAC) layers techniques, signal processing, spectrum allocation, collision analysis, resource management, modulation classification, interference mitigation, beamforming, and propagation attenuation, highlighting shortcomings and future strategies. In addition, the existing surveys in Table 2 focus on providing a comprehensive review of the THz communication and are not focusing on using ML methods for the THz performance improvements or discussing the available platforms such as commercial, testbeds and publicly available THz simulators. This survey fills this gap by presenting a constructive and comprehensive review of the use of AI in THz communications and available platforms. This survey also identifies issues and presents future recommendations to help researchers quickly identify the problem domain.

Ref.	Year	Primary Focus	Main Topics Covered by Surveys and Tutorials
[9]	2019	THz modulators	Reviewed THz modulators, evaluated devices, metamaterials, and modulation techniques for THz.
[10]	2019	THz channel features	Emphasized the important characteristics of THz channels and recent progress in device technologies.
[11]	2019	THz communications	Overviewed the advancements in THz communications and discussed the key technologies encountered in THz.
[12]	2019	THz channel models	Extensively presented the progress on THz device development and propagation model.
[13]	2019	THz-band operation	Surveyed the challenges, opportunities, potential applications, and future directions for THz.
[14]	2020	THz applications	Presented an outlook on THz and localization applications.
[15]	2020	THz channel features	Examined the characteristics of the THz channel and highlighted challenges.
[16]	2020	THz MAC protocols	Surveyed MAC protocols designed for THz.
[17]	2020	THz antennas analysis	Conducted a thorough examination of THz antennas, including their research background and fundamental principles.
[18]	2021	THz and <i>mmWave</i>	Comprehensively compared the <i>mmWave</i> and THz bands, and highlighted potential applications in 6G.
[19]	2021	THz capacity analysis	Presented characteristics and capacity analysis of THz channels.
[20]	2021	THz channel modeling	Focused on channel models and recent research efforts on factors affecting THz commu- nication and discussed future research on THz channel modeling.
[21]	2021	THz technology potential	Examined various aspects of THz technology, discussed the potential, and researched THz technology towards commercialization.
[22]	2021	THz communications improvement	Examined intelligent reflecting surfaces (IRS) for improving THz communications and addressed active and passive beamforming challenges.
[23]	2021	THz communication	Presented an in-depth overview of how IRS and THz communications can work together to create a flexible and adaptable wireless communication environment.
[24]	2021	THz signal processing	Provided an in-depth overview of THz channels and summarized recent research efforts on modeling these channels, and highlighted challenges.

Table 2. A summary of existing general surveys and tutorials on THz from 2019 to May 2023.

Table	e 2.	Cont.
-------	------	-------

Ref.	Year	Primary Focus	Main Topics Covered by Surveys and Tutorials
[25]	2021	THz for nanonetworks	Outlined areas where THz nanonetworks could be utilized, highlighting the specific requirements these applications would have for the supporting nanonetworks, which can be used as a guide for designing the necessary communication and networking protocols.
[26]	2021	DL techniques	Explored and discussed DL techniques for future 6G PHY communications systems.
[27]	2022	THz communication ad- vancements	Summarized the advancements in THz communication over the past decade and identi- fied the major obstacles that must be addressed for designing THz wireless systems in the future, including those for 6G networks, as well as short-distance connections and wireless connections for fixed locations.
[28]	2022	THz communication	Discussed THz communication aspects, examined theoretical approaches to analyze and design THz transmissions for increased connectivity and security, and explored spectrum management strategies.
[29]	2022	THz Propagation	Presented a comprehensive overview of the propagation channel characteristics and modeling for current THz applications.
[30]	2022	THz communication sys- tems	Investigated the unaddressed issues and emerging research fields that require further exploration in the context of THz-band communication systems.
[31]	2022	THz-Empowered UAVs in 6G	Highlighted opportunities, challenges, and design strategies that influence the benefits of integrating THz-based features and UAV networks.
[32]	2022	THz channel characteristic	Provided in-depth review of THz channel characteristics, highlighted open issues, and future directions on THz.
[33]	2022	THz physical, link, and network layers	Covered THz spectrum management, antennas, and beamforming, as well as integrating other technologies that enable 6G in THz communication.
[34]	2022	THz antenna design	Provided an overview of how ML, FL, and AI are being used in antenna design.
[35]	2022	RL-based security and privacy	Reviewed reinforcement learning (RL)-based physical (PHY)-layer security techniques in THz.
[36]	2022	ML methods for THz	Investigated signal processing and ML methods for THz sensing.
[37]	2022	ML techniques	Presented the use of distributed ML for 6G networks including THz.
[38]	2023	ML for THz beamforming	Covered ML techniques applied to <i>mmWave</i> and THz for beam management.
[39]	2023	AI in THz for healthcare	Reviewed AI methods used to advance THz technology for cancer detection applications.

1.3. Contributions of the Survey

The contribution of this survey paper regarding existing surveys and tutorials is outlined as follows:

- 1. A systematic method to detail the different THz performance aspects that are improved using ML techniques. We will delve into key THz challenges and how they are addressed through ML applications.
- 2. This survey gives readers a general understanding of the state of the art and current trends in utilizing ML to address specific THz issues and challenges.
- 3. This survey provides an in-depth overview and analysis of the THz platforms, including commercial platforms, testbeds, and publicly available network simulators. Additionally, this survey presents future research prospects for advancing THz simulators and the application of ML to improve or evaluate THz performance and provides readers with an understanding of what still needs to be accomplished in THz using ML.

1.4. Structure of the Survey

Figure 2 illustrates the hierarchical structure of the survey, where Section 2 presents a detailed background on *mmWave* and THz along with THz applications and a comparative study of *mmWave* and THz. Section 3 presents the need for AI in THz, challenges, the state of the art using DL and RL with their potentials, and shortcomings. Section 4 provides an overview and analysis of THz platforms, including commercial, Testbeds, and publicly available network simulators. Section 5 suggests future strategies in improving THz communication using simulation and AI methods. Finally, the last Section 6 provides concluding remarks on this survey paper.



Figure 2. Structure of the survey, illustrating the hierarchical organization of the survey with the overall survey design.

2. Background on mmWave and Terahertz

This section briefly discusses the *mmWave* and THz technologies and presents a comparative study of both technologies.

2.1. mmWave Overview

mmWave is a promising 5G mobile cellular network technology. Using *mmWave* MIMO small cells can significantly push the limits of legacy networks [40]. In the context of 5G and beyond, *mmWave* is expected to play a critical role in meeting the ever-increasing demand for data-intensive applications such as video streaming, real-time gaming, and virtual reality experiences [41]. The short wavelengths of *mmWave* signals also enable the deployment of small-cell wireless networks that provide high-speed connectivity in densely populated urban areas. The *mmWave* band consists of frequencies between 30 and 300 GHz, experiencing significant attenuation due to oxygen absorption [42,43]. However, 35, 94, 140, and 220 GHz frequencies have less attenuation and can be used for long-distance communication [42,44].

2.2. Terahertz Overview

The THz band, a trillion cycles per second, is the frequency range between 0.1 to 10 THz [12,45–47]. The advancement of wireless communication systems in the 6G and beyond is anticipated to rely heavily on THz technology [48]. With its ability to provide high-speed data transfer rates and low latency, THz technology is well-suited for various applications, including virtual and augmented reality [49], high-definition video streaming, and autonomous vehicles [33]. Higher data rates of up to Tbps make THz appropriate for imaging and communication networks. The THz band has been exploited for imaging for nearly two decades. Its non-ionizing properties make it safe for medical imaging and security checks. Furthermore, we discuss the standardization efforts on THz communications.

2.2.1. Standardization Efforts on THz Communications

IEEE 802.15.3d is a standard for THz, which defines a MAC layer and physical layer (PHY) for switched point-to-point links. The MAC layer supports aggregation and block acknowledgment to improve efficiency at high data rates. The PHY layer operates in the sub-THz frequency range between 252 GHz and 325 GHz [50–52]. The IEEE 802.15.3d defines two PHY modes that can be used to transmit data at speeds of up to 100 gigabits per second (Gb/s) in the range between 2.16 GHz and 69.12 GHz.

The IEEE 802.15.3d had been developed for nine years, from 2008 to 2017 [50]. The IEEE 802.15.3 Task Group 3d, formed in May 2014, developed a standard for THz switched point-to-point connections and a standard for proximity links at 60 GHz. The THz switched point-

to-point standard was approved and published in 2017. Furthermore, the development of the IEEE 802.15.3d standard is a significant effort, providing a foundation for developing new THz-based products and services, such as high-speed data links, medical imaging systems, and security and surveillance systems. The standard is also expected to accelerate the development of THz technology and its adoption by industry.

2.2.2. Applications of THz

The applications of THz can be categorized into macroscale and nanoscale. These applications require high speeds, ranging from Gbps to Tbps, for outdoor and indoor [53]. Additionally, some applications require lower but still high speeds, such as Gbps, small-cell networks, wireless local area networks (WLAN) [54], and communication between vehicles and devices [55]. THz applications at the macroscale and nanoscale are highlighted in Figure 3 [56].



Figure 3. Terahertz band applications at the macroscale and nanoscale [12,56,57].

2.2.3. Macroscale THz Applications

Macroscale applications range from 1–10 m [58] with dimensions greater than 100 nanometer (nm) [58]. Macroscale applications require Tbps links, including ultrahigh-definition video and holographic conferencing, gaming, and VR. In conventional cellular networks, THz is suited for indoor applications or high-speed wireless backhaul small cells [57]. Similarly, in conventional WLAN applications, Terabit Wireless Local Area Networks (T-WLAN) can seamlessly connect high-speed wired networks, such as optical fiber links. In the same way, Terabit Wireless Personal Area Networks (T-WPAN) enable ultra-high-speed communication between nearby devices. For instance, kiosk downloading is one type of T-WPAN application where large video content is transferred to nearby devices [18,59]. Furthermore, high path loss and absorption (e.g., in indoor and outdoor scenarios) limit the distance of macroscale applications [60,61].

2.2.4. Nanoscale THz Applications

Nanoscale applications are short-range, usually less than 1 m. Specifically, nanoscale refers to dimensions of 1 to 100 nm as defined in the "IEEE Standard Data Model for Nanoscale Communication Systems" and "IEEE Recommended Practice for Nanoscale and Molecular Communication Framework" [58,62]. The transmission distance of THz waves can vary depending on the application [18]. For example, THz waves can be used to

transmit data over a few micrometers in medical imaging. However, it can also be used to transmit data over a few meters in the context of wireless communication [18].

A nanosensor network is formed by distributing several nanosensors around the body (e.g., for surgical procedures) for data collection, where every sensor is equipped with a carbon nanotube (CNT)-based nanoantenna, acting as a transceiver and communicates using the THz frequency band [58,63]. Nanotechnology allows the creation of such tiny, specialized nanocomponents that can perform specific tasks such as data storage, computation, and sensing [63–65]. These nanodevices can communicate with each other in a centralized or distributed manner, making them useful for biomedical, industrial, military, and health monitoring applications. As an example of nanoscale, sodium, glucose, and other ions in a body can be monitored [58]. THz communication can transmit the measured data from the human body to a device outside the body. These data can then be processed by an external device, such as a mobile phone or smart band, and sent to a medical device or doctor [18]. Other studies in [60,61] investigated the propagation in human tissues, focusing on fat absorption under the THz band. They revealed that fat absorbs THz waves strongly, limiting the range of communication between nanodevices in human tissues. However, the results also show that a distance of a few millimeters might be sufficient to ensure a reliable communication link.

2.3. Comparison between THz and mmWave

Table 3 briefly compares *mmWave* communication and THz band. The THz and *mmWave* bands are adjacent to each other, but they have distinct properties. For example, the THz band offers greater bandwidth than the *mmWave* band, allowing higher data rates and reduced interference [66–69].

Table 3. Comparison between mmWave and THz wireless communication technologies [16].

Features	mmWave	Terahertz
Transceivers Device	Available [70]	Time-domain spectroscopy (TDS) and photonic-based frequency-domain spectroscopy (FDS) [71–74]
Modulation	High order modulation, supports 1024-QAM for	low order modulation [57], supports Time Spread On-
	downlink and 256-QAM for sub-7 GHz uplink	Off Keying (TS-OOK), Rate Division (RD-TS-OOK), Di-
	[75–77]	rect Sequence (DS-OOK) [78]
Antenna	Omni, directional, MIMO supported with high gain [79]	Omni and directional with phased array [75,80]
Bandwidth	0.03–0.3 THz [38]	0.1–10 THz [38]
Wavelength [mm] [55]	3–10	0.03–3
Array size [55]	10 imes 10	100×100
Channel models [81]	Available [82–87]	Available [88–93]
Standards	5G NR [94], IEEE 802.11ad [95], and IEEE	IEEE 802.15.3d [50]
	802.11ay [96]	
Mobility	Yes [97]	Yes [98,99]
Beam Management	Yes [100]	Yes [38]
Outdoor Deployment	Yes [79,84]	Yes [90]
Free-space loss	Low	High
Coverage	High [101]	Low
Range	Up to 200 m or less [102,103]	Up to 10 m [18,57]
Device size	A few millimeters	1 to 100 nanometer [58,62]
Wavelength	1 mm–30 μm [104,105]	3 cm–1 mm
Data rate	Up to 10 Gbps [59,79]	Up to 100 Gbps [50]
Power consumption	Medium	Medium
Network topology	Centralized and clustered	Centralized, clustered, and distributed
Line-of-sight	Both	Both
Non-line-of-sight	Both	Both
Source of noise	Thermal and molecular noise	Thermal
Simulators	5G cellular network [106], 5G NR networks [107]	TeraMIMO [108], NYUSIM [109], CloudRT [110], Nano-
		Sim [111], THz propagation [112], TeraSim [113],
		TeraSim-6G [114], and TeraSim-MAC [115]

The THz band has transmission windows that vary with distance, reaching up to THz bandwidth, while the *mmWave* band has a 10 GHz bandwidth and cannot achieve Tbps link speeds. For THz to attain data rates of up to 100 Gbps, the link capacity must be several times greater than the necessary user data rate to ensure prompt data delivery [59,116]. As frequencies increase in the THz band, Tbps links can achieve moderate and practical spectral efficiencies of a few bits per second per hertz. The THz band permits higher link directionality than *mmWave* at the same transmitter aperture due to less free-space diffraction and shorter wavelengths than *mmWave*. Smaller antennas with strong directivity in THz communications can also decrease transmitted power and interference. Additionally, the risk of eavesdropping is lower in THz bands than in *mmWave* bands because of THz high directionality beams [42,93,117,118].

3. Terahertz Meets AI

This section highlights state-of-the-art THz communications, including the need for AI in THz technology, challenges, state-of-the-art deep learning and reinforcement learning solutions, their potential, and shortcomings.

3.1. Need for AI in THz

AI comprises various techniques, such as ML, DL, and RL, that can be used to analyze and interpret large complex datasets. In recent years, the use of AI in THz has been gaining momentum as it is becoming increasingly important. Furthermore, the demand for faster and more efficient communication and sensing technologies continues to grow. Therefore, AI with reduced complexity is recommended in model and algorithm deficit cases [119]. The lack of a proper model is called the "model deficit," which can be caused by a lack of knowledge about the specific domain.

On the other hand, the "algorithm deficit" refers to a situation where a mathematical model is appropriately known. However, the optimization of such mathematical models is considered challenging. Moreover, AI methods seamlessly integrate contextual data into the decision-making process. Thus, AI can optimize the performance of THz communication systems by adaptively adjusting the system parameters to account for changing channel conditions, THz sensing, advanced signal processing techniques, and so on [120].

3.2. Challenges

THz band has the potential to provide high data rates and low latency. However, it faces several challenges, including limiting the distance due to high path loss, propagation attenuation, and absorption in indoor and outdoor environments [60,61], the narrow bandwidth of the THz band [121], channel estimation and beam training in multi-hop THz communications for intelligent MAC design [98,122], resource allocation in D2D scenarios [123,124], and modulation schemes classifications. Researchers have used DL and RL methods to address these challenges, as discussed in Sections 3.3 and 3.4, respectively. Additionally, the potentials and shortcomings of these methods are highlighted in Tables 4 and 5.

3.3. Deep Learning: State of the Art

The emergence of DL in the THz is a cutting-edge and rapidly expanding field that holds tremendous promise for revolutionizing THz research and transforming its applications. Table 4 illustrates DL methods applied to improve the THz technology with their potential and shortcomings. The highlighted approaches in Table 4 are designed to improve beam prediction [125], overcome noise reduction [126], improve spectrum allocation [127], classify THz modulation [128], to analyze collision and overcome it by designing an intelligent MAC [98], identify the type of modulation used in THz communication system [129], improve resource allocation [124], and classify THz bandwidth [121].

The beam misalignment problem caused due to the mobility of UAVs [130], was investigated and resolved by the authors in [131] by proposing RNN based on echo state

networks to reduce the outage probability of THz-band wireless links in UAV networks. Their proposed LeTera method uses a UAV mobility pattern to predict the optimal beam width. The LeTera was evaluated in various weather conditions under UAV flight experiments. The results of the LeTera method showed 99% accuracy, showing LeTera predicts the optimal beam width accurately. Furthermore, it was suggested that LeTera could be used to reduce outages and improve link capacity.

The paper [127] proposed an unsupervised learning-based approach for THz spectrum allocation in a multiuser THz network. Their proposed method divided the THz spectrum into sub-bands with unequal bandwidths to determine the optimal sub-bandwidth and transmit power. They trained a DNN model and approximated the near-optimal solutions. The numerical results illustrated that their proposed DNN method achieved a higher data rate than existing approaches.

The authors [128] proposed a convolutional neural network (CNN) approach, which uses the constellation diagram of the received signal as the input and applies a two-stage CNN architecture for modulation recognition. The first stage applies a convolution layer to extract the features, and the second stage uses fully connected layers for classification. The proposed approach was compared with other existing methods, and simulation results showed that it outperforms traditional techniques, especially for low signal-to-noise ratio (SNR) values.

Ref.	Year	DL Method	Potentials	Shortcomings
[131]	2020	RNN	Efficient beam alignment in UAVs	The LeTera scheme may not be effective in environments with obsta- cles since obstacles can block the beam and cause outages.
[125]	2021	GRU	Enhances beam prediction	It is dependent on a large training dataset for predicting the beam- forming weights. It may limit the range and reliability of beam prediction even with advanced machine learning models due to the high attenuation of THz signals.
[126]	2021	DNN	Reduces noise	DNN model can result in poor performance when the model is applied to new THz signals with different types or levels of noise.
[127]	2022	DNN	Efficiently allocates spec- trum	The performance of the DNN-based approach for spectrum allocation can be affected by the complexity of the wireless environment.
[128]	2022	CNN	Effectively classifies modu- lation	CNN is dependent on a large training dataset for accurately clas- sifying different modulation schemes, which can be challenging to obtain in THz communication systems due to the limited availability of hardware.
[98]	2022	LSTM, GRU, and Bi-LSTM	Reduces collision of packets	The dataset collection is not described in detail, which limits the regener- ation of the collision analysis through LSTM, Bi-LSTM, and GRU.
[129]	2022	CNN and LSTM	Recognizes signal modula- tion effectively	The approach may require large labeled data for training to achieve high accuracy, which could be a challenge to obtain in some in- door/outdoor scenarios.
[124]	2023	LSTM	Efficient resource manage- ment	Relies on simulation results rather than actual experimental data, which may limit the generalization of the findings to real-world scenarios.
[121]	2023	CNN	Classifies modulation and bandwidth	The experiments were conducted at a single frequency (120 GHz) and may not generalize to other frequencies in the THz band.
[65]	2023	CNN & GA	Efficient beam management at THz frequencies	Genetic algorithm is a heuristic algorithm, which may not guarantee the best solution.

Table 4. DL examples applied to THz technology.

Ref.	Year	DL method	Potentials	Shortcomings
[132]	2020	DRL	Reduces propagation atten- uation	DRL requires large amounts of training data to learn the underly- ing propagation environment, which can be time-consuming and resource-intensive.
[133]	2020	RL	Mitigates interference	The proposed approach assumes that interference is sporadic and short-lived, which may not be the case in all scenarios and could limit the performance of the algorithm.
[134]	2021	MAB	Efficient resource allocation in D2D scenario under multi-hop communication	RL methods can be computationally expensive and may require significant resources to implement.
[135]	2022	FDRL	Improved throughput using beamforming	Only SINR feedback as CSI may not perform real-world scenarios, which could limit its practical implementation.
[136]	2022	DRL-based FL	Efficient resource allocation to D2D-enabled wireless net- works	In a D2D environment, user mobility can vary widely. It may not conform to the assumed models, leading to sub-optimal resource allocation and reduced network efficiency.
[99]	2022	GAN	Improved routing in a dy- namic network using intel- ligent MAC protocol	It does not account for interference caused by the movement of nodes in other communication links. This can result in unexpected disruptions in communication links and impact the overall network performance.
[137]	2022	DQN	Minimizes network latency efficiently	It does not account for collaborative computation offloading among multiple UAVs and potential errors in channel estimation.
[138]	2023	RL	Enhances throughput of the network	The work does not consider the impact of noise and interference on the performance of the routing scheme.
[38]	2023	DRL	Improved energy efficiency	DRL is a computationally intensive technique, which may not be suitable for large-scale UDNs.

Table 5. RL examples applied to THz technology.

The paper [129] proposed a DL method to recognize signal modulation schemes at the THz frequency band. First, the paper highlighted the challenges of base-band signal processing in the THz frequency band due to the erratic fluctuations caused by severe weather and urban multipath scattering. Next, the paper explored DL techniques to address this challenge, specifically CNN and long short-term memory (LSTM) networks, to recognize signal modulation schemes. Finally, their study established indoor and outdoor environments to test the proposed DL method in severe weather conditions and varying SNR situations. The results of the study demonstrated the efficacy of DL-based techniques for modulation recognition in the THz frequency band.

In [98], the authors enhanced the existing MAC protocol called Adaptive Directional Antenna Protocol for Terahertz (ADAPT) [115], using multi-layer recurrent neural network (RNN). The ADAPT protocol shown in [115] comprises a 3-way handshake scheme using a turning AP. After turning into a new sector, the AP transmits a "call to action (CTA)" control packet to nodes. The nodes answer with a "request to send (RTS)" packet to the AP if the node has data to send. The AP waits for T_{wait} time. To this end, there are three possible cases: (1) zero RTS received, (2) 1 RTS received, and (3) greater than 1 RTS received. In case (1), the sector has no nodes; as a result, the AP skips the current sector and turns into a new sector. Meanwhile, in case (2), the AP has received exactly one RTS from the node. After T_{wait} time, the AP sends "clear to send (CTS)," completing the 3-way handshake. After the 3-way handshake, data transmission occurs between the node and AP. Finally, in case (3), the AP has received more than 1 RTSs. In such a situation, the AP divides the transmission slot for nodes. However, only one data packet can be transmitted by the node to AP in each sector. The [98] analyzed the collision behavior of ADAPT protocol considering macroscale use case. In their simulation, the BS was placed in the middle of a cell comprising 18 m of radius, where the devices were distributed randomly (i.e., ranging from 15 to 960). Their study shows the collision rate of control frames for the two topologies (centered and uniform) and two population values considering 30 sectors and 960 mobile nodes. As the population density increases, the collision rate also increases. The [98] evaluated the efficiency of sector collision classification using different RNN types

(LSTM, Bi-LSTM, and GRU). The test accuracy of sector collisions of Bi-LSTM was higher (i.e., 90.03%) compared to LSTM (85.04%) and GRU (i.e., 87.66%).

A new resource management scheme for THz communication is proposed in [124]. The scheme jointly optimizes the reflection coefficient of reconfigurable intelligent surfaces (RISs), the transmit power of the base station (BS), and the wideband THz resource block allocation. RISs are a type of metasurface that can be used to reflect and redirect electromagnetic waves. This can be used to improve the performance of THz communication systems by reducing path loss and increasing the bandwidth. The method in [124] adopts an LSTM relying on optimization and ensemble learning methods. Simulation results showed significant spectral efficiency gains for eMBB while ensuring the reliability and latency requirements of the URLLC service. The proposed scheme is a promising approach for improving the performance of THz communication systems. It can support various services, including eMBB, URLLC, and virtual reality/augmented reality (VR/AR). However, the real-time performance of the ensemble learning model is slightly worse than the optimization approach.

The authors in [121] leveraged a DL method based on CNN, aiming to classify modulation and bandwidth at the THz band. The dataset was collected at 120 GHz for various MCS and SNR. The CNN model was trained on the dataset and achieved accuracies of up to 78% and 90%. They also proposed a boosting technique to improve inference quality while accounting for memory and latency constraints and evaluated the latency of the proposed DL method by deploying it using FPGA. Their study provided critical insights into the potential of DL at the PHY layer for adaptive THz communications.

A new method for designing metasurfaces at THz frequencies was proposed in [65]. Their method combines CNN and genetic algorithms (GA). First, they trained a CNN to predict the amplitude and phase response of a metasurface based on the provided pattern, allowing it to speed up the forward prediction process. Then, once the CNN is trained, the authors create an inverse design model based on GA for the metasurface patterns that meet the desired amplitude and phase requirements. At the time of testing, it was shown that a metasurface pattern split the input beam into two beams with uniform power distribution. The GA-based metasurface design model found this pattern in only 10 min, faster than the traditional trial-and-error method.

3.4. Reinforcement Learning: State of the Art

RL has the potential to revolutionize THz technology by enabling more efficient and effective use of THz radiation in a wide range of applications. Table 5 illustrates RL methods applied to improve the THz technology with their potential and shortcomings. The high-lighted approaches in Table 5 are designed to improve the propagation attenuation [132], interference mitigation [133], multi-hop communication [134], THz beamforming [135], resource allocation to D2D-enabled wireless networks [136], MAC protocol for LoS mobile networks [99], latency minimization [137] etc.

To address high propagation attenuation and molecular absorption, a hybrid beamforming scheme was proposed in a multi-hop environment [132]. They combined a digital beamforming matrix at the BS with analog beamforming matrices at the RIS. Their proposed method leverages DRL to optimize beamforming and combat signal loss. Simulation results showed that this approach increases the coverage range of THz communications by 50% compared to existing benchmarks.

The authors proposed an RL framework to mitigate intermittent interference in THz networks [133]. Their proposed RL framework uses an adaptive multi-thresholding strategy to mitigate interference from directional links in the time domain. Moreover, the approach does not depend on pre-existing knowledge of interference statistics, making it an appropriate solution for mitigating interference in ever-changing scenarios. Simulation outcomes validated the BER performance of the proposed technique when compared to conventional time-domain interference mitigation strategies.

To tackle the challenges associated with multi-hop THz communications, the authors [134] proposed a new approach to beam training based on RL. First, the authors of the paper divide the multi-hop THz link into several individual single-hop links. They then train an RL agent to dynamically and collaboratively select the number of beam training levels across all individual single-hop links. Their proposed RL solution, based on the multi-armed bandit (MAB) framework, demonstrated rapid convergence in simulations with random channels and noise. Furthermore, they showed that their approach achieves a significant performance gain over traditional hierarchical beam training with a fixed number of training levels. This new approach to beam training is a promising step towards realizing multi-hop THz communications. It is a more efficient and effective way to train beams and improve the performance of multi-hop THz links.

The paper [135] introduced federated learning (FL) approach to training the beam. First, the authors divided the cellular network into several base stations. They then trained a deep deterministic policy gradient (DDPG) agent at each BS to learn a THz beamforming policy with limited channel state information (CSI). Moreover, they update their DDPG models using hidden information to reduce inter-cell interference. The hidden information, which is interference from the estimated CSI, is extracted and exchanged by the federated edge learning (FEL) server. Simulation results indicated that their approach achieves a significant performance gain over conventional non-learning-based and existing non-FDRL benchmark optimization techniques. Additionally, simulation results indicated that the cellular network could achieve higher throughput when more THz CSI and hidden neurons of DDPG are incorporated.

The paper [136] proposed a resource allocation (e.g., mode selection, power control, and channel allocation) approach using DRL and FL. They then trained a DRL agent at each cell to learn how to allocate resources to maximize the overall capacity and minimize power consumption while guaranteeing the quality of service (QoS) requirement of both cellular and D2D users. The DRL agents are updated using FL, which allows them to learn from each other without sharing their data. Their simulation results indicated that the proposed approach is robust to channel variations.

In [99], the authors proposed a MAC algorithm for mobile THz airborne networks (TANs). Their proposed protocol comprises three distinct characteristics: spatiotemporal TAN state learning, precise two-tiered MAC operational control, and comprehensive TAN-specific MAC behavior management. The primary aim was to create a predictive network state estimation model using deep learning and generative adversarial networks (GANs), enabling the coordination of all one-hop neighbors for scheduled, antenna-aligned line-of-sight communication. In addition, the authors suggested implementing nested deep reinforcement learning (DRL) featuring outer and inner policy loops for determining high-level and low-level actions. Simulation results revealed the seamless, high-speed THz communication at the MAC layer, supported by robust RF links under mobility scenarios.

A DRL algorithm for joint optimization of UAV placement, resource allocation, and computation offloading is investigated [137]. The proposed deep Q-learning (DQN) and DDPG search for near-optimal solutions were studied in a highly dynamic environment. Simulation results in different scenarios demonstrated the effectiveness of the proposed algorithms in solving the formulated non-convex problem of minimizing latency. The proposed solution is a promising approach, increasing the system capacity and meeting the real-time demands of latency-sensitive applications.

An RL analytical model was proposed for a multi-hop scenario in [138], inspired by the human cardiovascular system based on the use of Markov decision processes (MDPs). Their proposed MDP model is a four-dimensional (4D), simultaneously characterizes the nanodevice motion in the bloodstream and its energy level. The model was evaluated using MATLAB and *Simpy* simulators considering a hand vein scenario. It was found that multi-hop scenarios can significantly improve the throughput of the nanonetwork without sharply penalizing other aspects, such as energy consumption. Furthermore, their study suggests that RL-based dynamic multi-hop routing schemes can be a valuable tool for flow-guided nanonetworks in the human cardiovascular system.

The authors in [38] proposed a decentralized DRL-based technique to maximize energy efficiency in ultra-dense networks. Their proposed DRL method comprises two networks: actor and critic. Actor-network is responsible for determining the node association using the local data. In contrast, a critic network determines the energy-efficient user association and informs the decision of the actor-network. In their proposed DRL method, the DRL agent in each base station can determine the user association decision. Their results showed that their method achieved significant energy efficiency gains (i.e., more than 50%) over conventional techniques. Furthermore, it was suggested that the method could be utilized for user scheduling and resource allocation.

4. Terahertz Platforms

To quickly progress in communication and networking solutions, it is essential to develop experimental testbeds and simulation tools in parallel [139–144]. The THz technology is rapidly evolving, and various commercial, testbed, and simulator platforms are available to researchers and developers. The remainder of this section discusses the available platform in detail.

4.1. THz Commercial Platforms

Typically, these platforms offer the most features and performance. Therefore, they are a good option for researchers and developers who need a turnkey solution that can be used to develop and deploy THz applications quickly and easily.

There are two types of commercial THz devices: time-domain spectroscopy (TDS) and photonic-based frequency-domain spectroscopy (FDS), a widely used method for THz measurements [71–74,145]. TDS devices use a THz pulse to excite a material and measure the time it takes for the material to respond. The time-domain response is then converted to a frequency-domain spectrum. TDS devices are non-destructive and can measure a wide range of materials. However, they can be time-consuming and require high-power THz sources. On the other hand, FDS is a continuous-wave and frequency-tunable method, producing narrow-band signals. These devices use a THz field to excite a material and measure the frequency-domain response of the material. FDS devices are fast and can measure a wide range of materials. However, they are sensitive to environmental factors and require high-quality optical components.

Various platforms are available from different vendors, offering different features and capabilities. As an example of THz platforms, such as ZEMAX THz Design Studio [146], TERAVIEW THz-3000 [147], Menlo Systems THz-QCL source [148], and Advantest THz-3000 [149] are available. A comparison table of these commercial platforms is illustrated in Table 6.

Table 6. Comparison of commercial platforms for THz communications considering various features.

Device	Manufacturer	Frequency Range	Wavelength Range	Applications
ZEMAX THz Design Studio	ZEMAX	0.1–10 THz	300 µm–3 mm	design and simulation of THz components and systems
TERAVIEW THz-3000	TeraView	0.1–3 THz	3 mm–100 μm	spectroscopy, imaging, sensing
Menlo Systems THz-QCL	Menlo Systems	0.1–10 THz	300 µm–3 mm	spectroscopy, imaging, sensing, communications
Advantest THz-3000	Advantest	1–10 THz	300 µm–3 mm	material inspection and analysis

4.2. THz Testbeds

A testbed is a platform for testing and validating new technologies and techniques in a controlled and reproducible environment. Testbeds are essential for research and development in THz communication, allowing researchers to test and optimize the designs before deploying them in real-world scenarios. Recently, the TeraNova testbed was utilized in [143], offering a wide range of features, including point-to-point links, multi-hop networks, and free-space optical communications, making it ideal for the development and evaluation of new THz communication technologies. The TeraNova testbed is a joint project between the University of New South Wales in Australia and the University of California, Berkeley in the United States. The TeraNova platform is an integrated testbed for ultrabroadband wireless communications at THz frequencies [143]. The TeraNova testbed platform primarily comprises a transmitter and a receiver, operating at frequencies up to 1.05 THz. The transmitter is based on a Schottky-diode [150] frequency multiplying and mixing chain, and the receiver is based on a sub-harmonic frequency mixer. The transmitter and the receiver are integrated with digital signal processing back-ends that allow various modulation and coding schemes to be implemented.

4.3. THz Simulators

A simulator is an essential tool for analyzing and evaluating the performance of wireless communication systems. With the ever-increasing demand for high-speed and reliable communication networks, simulation tools have become even more critical in designing and optimizing these systems. In the context of THz communication, where experimental setups are often costly and challenging, simulations are even more valuable. Furthermore, simulation tools can help to validate and improve solutions by providing a safe and controlled environment. Here, we present several publicly available THz network simulators, TeraMIMO [108], NYUSIM [109,151,152], CloudRT [32,110], Nano-Sim [111], THz propagation [112], TeraSim [113], TeraSim-6G [114], and TeraSim-MAC [115].

4.3.1. TeraMIMO

The TeraMIMO [108] channel simulator was designed in MATLAB for wideband ultramassive MIMO THz communications. The simulator employs a Monte Carlo-based method to generate random channel realizations with realistic fading and shadowing effects by modeling the THz channel in both the time and frequency domains. It includes directional antenna patterns, beamforming, and spatial correlation features. TeraMIMO supports two modes of operations: (1) graphical user interface (GUI) and (2) MATLAB scripting. In the GUI mode, the user can input the simulation parameters (e.g., general, UM-MIMO, and THz-specific parameters) and then run the simulation. After the simulation is executed, the results are stored in the workspace. Overall, the TeraMIMO channel simulator is a powerful tool for researchers and engineers developing THz communication systems. It allows system performance evaluation in realistic channel conditions, and its modular design allows for the simulation of various scenarios. The ability to handle UM-MIMO systems and wideband signals makes it particularly useful for evaluating the potential of THz communications for future high-speed wireless networks.

4.3.2. NYUSIM

The New York University (NYU) developed a simulator called NYUSIM based on their measurement for *mmWave* and THz channels [109,151]. The simulator was built on real-world measurements at multiple *mmWave* frequencies, from 28 to 73 GHz. NYUSIM 4.0 [152] has been expanded to the frequency range of 0.5–150 GHz and supports simulations for urban micro, urban macro, rural macro, and indoor hotspot scenarios [153].

4.3.3. CloudRT

CloudRT is a cloud-based platform designed for ray tracing that can be used to simulate the performance of wireless communication systems, such as 5G and THz, in a variety of applications, such as vehicle-to-infrastructure (V2I) [32,110]. CloudRT supports various frequencies, from 450 MHz to 325 GHz. It has been used to simulate the propagation of THz waves in various environments, including urban, rural, indoor/outdoor, and

railway [154–158]. The CloudRT is a powerful tool that can be used to design and optimize THz communication systems performance.

4.3.4. Nano-Sim

Nano-Sim is an ns-3 extension developed by the authors of [111] to model electromagnetic nanoscale communication networks in the THz range with a propagation loss model adapted from [159]. However, the THz-band communication scenario is only implemented at the nanoscale. Moreover, it utilizes a streamlined channel model with the nano-node transmission range as its only parameter. Furthermore, it fails to account for the need for nanodevices for energy harvesting.

4.3.5. THz Propagation

For frequencies up to 2 THz, the authors of [112] created a THz directional propagation loss model for ns-3. This model is used for the cases where both the transmitter and receiver use collimated lenses (e.g., light rays are parallel to each other), resulting in the signal in a minimum divergence. However, in the terahertz domain, atmospheric molecules cause signal attenuation. This module accurately predicts THz propagation loss for a single frequency. However, it is not a complete network simulator module since neither the PHY nor MAC layer is implemented.

4.3.6. TeraSim

The Ultrabroadband Nanonetworking Laboratory (UN-Lab) developed ns-3 module for THz band, namely "TeraSim". The simulator was designed for nanoscale and macroscale applications [113,115]. This ns-3 module can simulate THz-band communication networks, which operate in the frequency range of 0.1 to 10 THz [115,160]. TeraSim takes into account the unique characteristics of THz devices and the THz channel, including a standard channel module that uses a frequency-selective channel model, parallel modules for nanoscale and macroscale situations, a THz directional antenna model, and an energy harvesting model [161–163]. In addition, TeraSim implements PHY and MAC layers modules, tailored with well-known MAC protocols ALOHA and CSMA at the link layer.

4.3.7. TeraSim-6G

TeraSim-6G was introduced in [114], which extends upon the previously developed TeraSim simulator [113]. The TeraSim-6G comprises two parts: MATLAB and ns-3. The TeraSim-6G implements the channel models from [82,161] in MATLAB. The MATLAB part is then integrated with the ns-3 module for a realistic simulation environment. The ns-3 part contains the *mmWave* and THz scripts for various scenarios, such as short and long distances, with coverage of 1 to 5 m and 10 to 20 m, respectively. TeraSim-6G considers a THz link operating at 1.0345 THz with 74 GHz of bandwidth and a *mmWave* link operating at 28 GHz with the maximum bandwidth allowed in 3GPP NR (e.g., 400 MHz) using a constant bit rate source with UDP at the transport layer.

4.3.8. TeraSim-MAC

TeraSim-MAC [115] is the extended version of TeraSim ns-3 module [113], which describes the implementation of a new MAC protocol for the macroscale scenario. In addition, TeraSim-MAC models the peculiarities of the THz frequency-selective channel. In this module, the authors used IEEE 802.15.3d standard for high data rate wireless networks [50]. TeraSim-MAC uses a single carrier signal with a bandwidth of 69.12 GHz, operating in the frequency range of 252.72 to 321.84 GHz.

Comparative Analysis of the THz Simulators

A detailed comparison of the available THz ns-3 modules and other simulators is illustrated in Table 7. Among them, TeraSim is the widely adopted ns-3 simulator module in the literature due to open-source and the availability of basic THz features, including

physical layer protocol (e.g., IEEE 802.15.3d) and a wide range of frequency spectrum. Apart from it, these two works [114,115] extended the TeraSim by adding exciting new THz features along the acknowledgment (ACK) support and MAC protocols.

Features	Nano-Sim	THz Prop- agation	TeraSim	TeraSim-6G	TeraSim- MAC	TeraMIMO	NYUSIM	CloudRT
Simulation platform	ns-3	ns-3	ns-3	ns-3	ns-3	MATLAB	standalone	standalone
Application scenario	nano	nano	nano & macro	nano	nano & macro	macro	macro	macro
Physical	X	X	X	X	IEEE 802 15 3d	X	X	X
Propagation loss model	\checkmark	\checkmark	\checkmark	\checkmark	√	\checkmark	X	X
Energy harvest- ing model	X	X	\checkmark	\checkmark	\checkmark	X	X	X
Mobility	X	X	X	X	X	X	X	\checkmark
Adaptive MCS	X	X	X	X	\checkmark	X	X	X

Table 7. Comparison of available THz simulator modules and standalone simulators.

5. Future Strategies

The future strategies are broadly categorized into simulator enhancements and using ML techniques to evaluate and potentially improve THz performance.

5.1. Simulator Enhancements

Current THz simulators offer a promising set of features, but they can be further enhanced to provide more realistic evaluations of the performance of THz systems. This can be achieved by adding mobility, downlink traffic, multiple antennas, and building environments. Therefore, it will allow the researchers to understand better how THz systems will perform in real-world scenarios, which will help to design more efficient and effective MAC protocols for THz system.

5.1.1. Mobility

Mobility is one of the essential aspects of communication, where nodes can move from one location to another. In the 6G communications system, some applications require mobility in indoor and outdoor scenarios, for instance, tracking applications [55]. Thus, it would be beneficial to introduce node mobility in *TeraSim-MAC* ns-3 module to evaluate the behavior and performance of the THz system. The mobility models available in ns-3 include Constant Position, Constant Velocity, Constant Acceleration, Gaussian Markov, Hierarchical, Random Direction 2D, Random Walk 2D, Random Waypoint, Steady State Random Waypoint, and Waypoint [164]. The most common and widely adopted mobility model in the literature is the Random Walk 2D mobility model for asset tracking [165,166].

5.1.2. Downlink Traffic

Downlink traffic refers to data transmission from a central source (access point or network server) to multiple receivers (mobile devices or other networked devices). THz communication poses challenges regarding signal propagation, as THz frequencies are affected by atmospheric absorption and scattering. Therefore, efficiently managing downlink traffic in a THz communication system will require advanced techniques such as beamforming [167], modulation, and error correction.

5.1.3. Multiple Antennas in AP

Using multiple antennas in AP can significantly improve the performance of wireless communication systems. MIMO can be used to increase the capacity, coverage, and reliability of THz-based wireless networks. However, several issues must be addressed when implementing multiple antennas in THz APs. One of the main issues is the high cost

of THz equipment, which can make it challenging to deploy large antennas. Additionally, the limited range of THz waves makes it difficult to achieve a wide area with a single AP.

AI-enabled solutions can be used to overcome these issues and optimize the performance of multiple antennas in THz APs. For example, AI-based beamforming techniques can direct the signal toward the intended user and reduce interference. Additionally, AI-based algorithms can be used to optimize the deployment of multiple antennas, considering factors such as the location of users, obstacles, and other environmental factors. AI-based techniques can also be used to dynamically adjust the transmission power and modulation scheme of the APs to adapt to changing channel conditions and improve the overall performance of the wireless network.

5.1.4. Buildings

Building features can be introduced for a more realistic scenario for THz performance evaluation, such as scalability analysis in high-density urban areas, similar to [168]. NS-3 incorporates square-shaped buildings with dimensions and distances based on the Manhattan layout model, which includes correlated shadowing. Nodes within a specific building are classified as indoor, and their transmissions will encounter significant building penetration losses. The building parameters used for the urban environment in ns-3 are detailed in [169]. These parameters can be adopted in *TeraSim-MAC* to test the THz performance in a realistic environment.

5.2. ML-Based Enhancements

ML methods, such as DL, RL, and FL can be used to enhance the performance of THz in terms of interference, energy consumption, latency, and collision [170,171]. ML-based solutions will lead to the development of new THz systems that are more efficient, effective, and reliable.

5.2.1. Deep Learning

THz research primarily concentrates on devices, channels, and the PHY layer. Only a few MAC protocols for the THz band have been surveyed in [16]. A recent study in [98] proposed a DL-enhanced MAC protocol for THz communications. The proposed protocol in [98] utilizes the TeraSim-MAC simulator module to improve the collision of data packets. The results of their study show that the proposed protocol can significantly reduce the number of collisions in THz networks, therefore improving the overall performance of the network. Furthermore, incorporating DL techniques would be valuable when designing MAC protocols for THz communications using the ns-3 module TeraSim-MAC [113]. DL can be used to learn the most active sectors, dynamically adjust beamwidth, and select the most efficient modulation and coding scheme (MCS). This will lead to significant performance improvements in THz networks in terms of energy consumption.

A centralized DL approach is one example of using DL in the THz network. Generally, a centralized DL approach involves offline training and online observation processing for efficient radio resource allocation (e.g., MCS) to nodes [172,173]. The offline mode is mainly responsible for data collection (e.g., CSI features), cleaning, and training the DL model. After the DL model training, the pre-trained (i.e., inference) model is deployed on a network server or device, as illustrated in Figure 4. The inference model assigns resources based on the raw data generated during the network.

For instance, to use the inference model in ns-3, an open ecosystem called Open Neural Network Exchange (ONNX) [174] can be used as it is an open-source format for DL and traditional ML models. First, the ONNX-supported model can be generated using PyTorch. Then the pre-trained model (i.e., inference model) can be imported in ns-3 with the help of ONNX API for resource management by providing raw data during simulation.



Figure 4. Example of centralized deep learning: in centralized deep learning, the ML model is trained on a centralized server using a one-time generated offline dataset. The trained model (i.e., pre-trained or inference model) is then deployed on the (1) devices or (2) the edge server for efficient resource allocation (e.g., transmit power and MCS).

5.2.2. Federated Learning

FL is one the distributed machine learning methods that have been studied in various aspects of wireless communications [175–178]. Therefore, FL is considered one of the promising ML techniques for 6G communications [37,179–182], which can be utilized for THz in different ways, such as beamforming [38]. FL can be used to train a beamforming model optimized for the specific environment. This can improve signal strength and reduce interference between users. As an example, the authors in [183] have utilized FL for beamforming under ultra-dense *mmWave* networks. Therefore, the systematic design and performance analysis of beam management in FL-based ultra-dense THz networks is still open problems. Generally, FL trains an ML algorithm on decentralized devices without sharing the local data with a central server, as shown in Figure 5. The goal of FL is to improve the performance of ML models while protecting the privacy of user data.



Figure 5. Example of federated learning: the global model is trained on a central server, and the local models are trained on the devices. The edge server and the devices share their models periodically to improve the global model.

For example, FL can be used in ns-3 using the ns-3-FL simulator [184,185]. The ns-3-FL simulator provides several features, including the FL algorithm, sharing global and local FL models between the devices and the central server, training the local models, and aggregating the local models to form a global model. Such features can be integrated with the TeraSim-MAC simulator to implement FL beamforming and modulation techniques.

5.2.3. Reinforcement Learning

RL can enhance THz performance by learning from the interactions between the system and its surroundings. Applying RL for THz in ns-3 can provide a powerful tool for researchers and engineers to evaluate and optimize the performance of THz communication systems in a simulated environment.

One way to apply RL for THz in ns-3 is to optimize the parameters of a THz communication system, such as transmit power, modulation [186], and beamforming, based on the current channel conditions. For example, the RL algorithm can learn from past interactions with the simulated environment and adjust the system parameters to maximize a particular performance metric, such as data rate or capacity.

For instance, two widely adopted tools are available for RL in ns-3, such as ns3-gym [187,188] and ns3-AI [189,190].

- 1. *ns3-gym:* ns3-gym is a framework based on OpenAI Gym [191], allowing researchers from academia and developers from industry to use the ns-3 in conjunction with RL algorithms. Figure 6 illustrates the ns3-gym architecture comprising two primary components: the ns-3 and the OpenAI Gym framework. The former component is utilized for implementing environments, whereas the latter component is leveraged to unify the interface of these environments. Together, these components facilitate the development and deployment of robust and scalable RL algorithms for various networking applications. The authors in [188] utilized ns3-gym for radio channel selection in IEEE 802.11 network.
- *ns3-AI*: The ns3-AI architecture is illustrated in Figure 7 [190], enabling data interaction between ns-3 and other Python-based AI frameworks (e.g., TensorFlow and PyTorch). The authors in [190] utilized the ns3-AI for (a) channel quality indicator (CQI) prediction in NR using LSTM and (b) TCP congestion control.



Figure 6. The architecture of ns3-gym utilized in ns-3 for RL [188].

5.2.4. THz for Indoor Localization

Current indoor localization systems rely on conventional wireless technologies such as Wi-Fi [192], Bluetooth Low Energy (BLE) [193], Ultra-Wideband (UWB) [194], Inertial Measurement Units (IMUs) [195], and Geomagnetic Field Based [196], which are often limited by low accuracy, high multipath propagation, and non-line-of-sight conditions. However, THz communication, which operates at a higher frequency (frequency range of 0.1 to 10 THz [12]) than conventional wireless technologies, has shown the potential to provide high accuracy with a mean distance error of 0.27 m [197] and 0.25 m [198].

The performance of these methods [197,198] can be further enhanced by implementing a multi-modal system, such as Wi-Fi, to generate a radio fingerprinting map with the THz technology. This will involve collecting RSSI data from the THz and Wi-Fi modules and processing it to create a more accurate radio fingerprinting map. Training the DNN model with the newly generated dataset from the multi-modal system and testing in the online phase for localization will involve using the trained DNN model to estimate the location of a mobile device based on the radio fingerprinting map generated from the THz and Wi-Fi modules.



Figure 7. The architecture of ns3-AI utilized in ns-3 for reinforcement and DL [190].

6. Conclusions

THz is a promising technology for high-speed future wireless communication. However, many challenges need to be addressed before THz can be widely deployed, such as the high attenuation of THz signals in the atmosphere, beamforming, and medium access control layer protocols. One promising approach to overcoming these challenges is using AI, which can be used to develop new THz communication protocols that are more robust and efficient. Furthermore, the field of AI in THz is rapidly growing, with recent research and development focusing on various areas such as spectrum allocation, modulation and bandwidth classification, collision analysis, resource management, interference mitigation, intelligent beamforming, and intelligent medium access control layer protocols. Therefore, we expect more research and development, leading to new AI-based THz systems and applications.

In conclusion, this survey provides a detailed analysis of the state-of-the-art THz communications using AI. It discusses the available platforms, including commercial, testbeds, and publicly available network simulators. Finally, recommendations for THz are highlighted, which serve as a roadmap for improving the simulators and realizing the use of AI in THz.

Author Contributions: Conceptualization, A.F. and J.-Y.P.; methodology, A.F.; software, A.F.; validation, A.F. and J.-Y.P.; investigation, A.F. and J.-Y.P.; resources, A.F. and J.-Y.P.; data curation, A.F. and J.-Y.P.; writing—original draft preparation, A.F.; writing—review and editing, A.F. and J.-Y.P.; visualization, A.F.; supervision, A.F. and J.-Y.P.; project administration, A.F. and J.-Y.P.; funding acquisition, A.F. and J.-Y.P. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by Research Fund from Chosun University, 2022.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

21 of 29

Abbreviations

The following abbreviations are used in this manuscript:

3GPP	3rd generation partnership project
ACK	Acknowledgment
ADAPT	Adaptive Directional Antenna Protocol for Terahertz
AP	Access Point
BFR	Bit Frror Rate
BLF	Bluetooth Low Energy
BS	Base Station
	Code-Division Multiple Access
CNN	Convolutional Neural Network
COL	Channel Quality Indicator
CSI	Channel State Information
CTA	Call to Action
	Doop Deterministic Policy Cradient
DI	Deep Learning
DNN	Deep Neural Network
DON	Deep O Loarning
DQN	Deep Q-Leanning
EDCE	Enhanced Data Pates for CSM Evolution
oMBB	Enhanced Data Rates for G5W Evolution
	Effective Redicted Dever
	Enecure Radiated Fower
FDMA	Frequency-Division Multiple Access
FDKL	Frequency Demain Spectroscopy
FD5	Constructions A deconstruction National
GAN	Generative Adversarial Network
GPK5	General Packet Radio Service
GRU	Gated Recurrent Unit
GSIM	Global System for Mobile Communications
IMUS	Inertial Measurement Units
InH	Indoor Hotspot
101	Internet of Things
IKS	Intelligent Reflecting Surfaces
	International Telecommunication Union
	Long Short-Term Memory
	Long-Term Evolution
MAC	Medium Access Control
MBC	Modulation and Bandwidth Classification
MCS	Modulation and Coding Scheme
MIMO	Multiple-Input and Multiple-Output
mMIC	Massive Machine Type Communications
mmWave	Millimeter-Wave
NK	
NYUSIM	New York University Simulator
OFDM	Orthogonal Frequency-Division Multiplexing
ONNX	Open Neural Network Exchange
PHY	Physical Layer
QoS	Quality of Service
RL	Reinforcement Learning
RMa	Rural Macro
RTS	Request to Send
TAN	THz Airborne Network
TCP	Transmission Control Protocol
TDMA	Time-Division Multiple Access
TDS	Time-Domain Spectroscopy
TSN	Time Sensitive Networking
UAV	Unmanned Aerial Vehicle

UMa	Urban Macro
UMi	Urban Micro
UN-Lab	Ultrabroadband Nanonetworking Laboratory
URLLC	Ultra-Reliable Low-Latency Communications
UWB	Ultra-Wideband
V2I	Vehicle-to-Infrastructure
V2X	Vehicle-to-Everything
VR	Virtual Reality
	-

References

- Brilhante, D.d.S.; Manjarres, J.C.; Moreira, R.; de Oliveira Veiga, L.; de Rezende, J.F.; Müller, F.; Klautau, A.; Leonel Mendes, L.; de Figueiredo, P.F.A. A Literature Survey on AI-Aided Beamforming and Beam Management for 5G and 6G Systems. *Sensors* 2023, 23, 4359. [CrossRef] [PubMed]
- GSMA-3GPP; Technical Specification Group Radio Access Network; 5G; NR; NR and NG-RAN Overall Description; Stage-2 (3GPP TS 38.300 version 16.2.0 Release 16), 2020-07. Available online: https://www.etsi.org/deliver/etsi_ts/138300_138399/13 8300/16.02.00_60/ts_138300v160200p.pdf (accessed on 25 April 2023).
- 3. Chen, W.; Montojo, J.; Lee, J.; Shafi, M.; Kim, Y. The standardization of 5G-Advanced in 3GPP. *IEEE Commun. Mag.* 2022, 60, 98–104. [CrossRef]
- Ateya, A.A.; Muthanna, A.; Makolkina, M.; Koucheryavy, A. Study of 5G Services Standardization: Specifications and Requirements. In Proceedings of the 10th International Congress on Ultra Modern Telecommunications and Control Systems and Workshops (ICUMT), Moscow, Russia, 5–9 November 2018; pp. 1–6. [CrossRef]
- Ghosh, A.; Maeder, A.; Baker, M.; Chandramouli, D. 5G evolution: A view on 5G cellular technology beyond 3GPP release 15. IEEE Access 2019, 7, 127639–127651. [CrossRef]
- 6. Saxena, N.; Rastogi, E.; Rastogi, A. 6G Use Cases, Requirements, and Metrics. In 6G Mobile Wireless Networks; Springer: Berlin/Heidelberg, Germany, 2021; pp. 7–24.
- An Introduction to 6G Flagship Program. Available online: https://www.eucnc.eu/wp-content/uploads/2021/06/About_6G_ Flagship.pdf (accessed on 10 February 2023).
- 8. Kitchenham, B.A.; Charters, S. *Guidelines for Performing Systematic Literature Reviews in Software Engineering*; Technical Report EBSE 2007-001. Durham University Joint Report; Keele University: Staffordshire, UK, 2007.
- 9. Ma, Z.; Geng, Z.; Fan, Z.; Liu, J.; Chen, H. Modulators for terahertz communication: the current state of the art. *Research* 2019, 2019, 6482975. [CrossRef] [PubMed]
- 10. Han, C.; Wu, Y.; Chen, Z.; Wang, X. Terahertz communications (TeraCom): Challenges and impact on 6G wireless systems. *arXiv* **2019**, arXiv:1912.06040.
- 11. Chen, Z.; Ma, X.; Zhang, B.; Zhang, Y.; Niu, Z.; Kuang, N.; Chen, W.; Li, L.; Li, S. A survey on terahertz communications. *China Commun.* **2019**, *16*, 1–35. [CrossRef]
- 12. Elayan, H.; Amin, O.; Shihada, B.; Shubair, R.M.; Alouini, M.S. Terahertz band: The last piece of RF spectrum puzzle for communication systems. *IEEE Open J. Commun. Soci.* 2019, *1*, 1–32. [CrossRef]
- 13. Huq, K.M.S.; Busari, S.A.; Rodriguez, J.; Frascolla, V.; Bazzi, W.; Sicker, D.C. Terahertz-Enabled Wireless System for Beyond-5G Ultra-Fast Networks: A Brief Survey. *IEEE Netw.* **2019**, *33*, 89–95. [CrossRef]
- 14. Sarieddeen, H.; Saeed, N.; Al-Naffouri, T.Y.; Alouini, M.S. Next Generation Terahertz Communications: A Rendezvous of Sensing, Imaging, and Localization. *IEEE Commun. Mag.* 2020, *58*, 69–75. [CrossRef]
- 15. Faisal, A.; Sarieddeen, H.; Dahrouj, H.; Al-Naffouri, T.Y.; Alouini, M.S. Ultramassive MIMO Systems at Terahertz Bands: Prospects and Challenges. *IEEE Vehic. Technol. Mag.* 2020, *15*, 33–42. [CrossRef]
- 16. Ghafoor, S.; Boujnah, N.; Rehmani, M.H.; Davy, A. MAC protocols for terahertz communication: A comprehensive survey. *IEEE Commun. Surv. Tutor.* **2020**, *22*, 2236–2282. [CrossRef]
- 17. He, Y.; Chen, Y.; Zhang, L.; Wong, S.W.; Chen, Z.N. An overview of terahertz antennas. *China Commun.* 2020, 17, 124–165. [CrossRef]
- 18. Tripathi, S.; Sabu, N.V.; Gupta, A.K.; Dhillon, H.S. Millimeter-wave and terahertz spectrum for 6G wireless. In 6G Mobile Wireless Networks; Springer: Berlin/Heidelberg, Germany, 2021; pp. 83–121.
- Guan, K.; He, D.; Ai, B.; Chen, Y.; Han, C.; Peng, B.; Zhong, Z.; Kürner, T. Channel Characterization and Capacity Analysis for THz Communication Enabled Smart Rail Mobility. *IEEE Trans. Vehic. Technol.* 2021, 70, 4065–4080. [CrossRef]
- Liu, S.; Yu, X.; Guo, R.; Tang, Y.; Zhao, Z. THz channel modeling: Consolidating the road to THz communications. *China Commun.* 2021, 18, 33–49. [CrossRef]
- 21. Solyman, A.A.; Elhaty, I.A. Potential key challenges for terahertz communication systems. *Int. J. Elect. Comp. Eng.* **2021**, *11*, 3403–3409. [CrossRef]
- Chen, Z.; Ning, B.; Han, C.; Tian, Z.; Li, S. Intelligent Reflecting Surface Assisted Terahertz Communications Toward 6G. *IEEE Wireless Commun.* 2021, 28, 110–117. [CrossRef]

- Chen, Z.; Ma, X.; Han, C.; Wen, Q. Towards intelligent reflecting surface empowered 6G terahertz communications: A survey. China Commun. 2021, 18, 93–119. [CrossRef]
- 24. Sarieddeen, H.; Alouini, M.S.; Al-Naffouri, T.Y. An Overview of Signal Processing Techniques for Terahertz Communications. *Proc. IEEE* 2021, 109, 1628–1665. [CrossRef]
- Lemic, F.; Abadal, S.; Tavernier, W.; Stroobant, P.; Colle, D.; Alarcón, E.; Marquez-Barja, J.; Famaey, J. Survey on Terahertz Nanocommunication and Networking: A Top-Down Perspective. *IEEE J. Selec. Areas Commun.* 2021, 39, 1506–1543. [CrossRef]
- 26. Zhang, S.; Liu, J.; Rodrigues, T.K.; Kato, N. Deep learning techniques for advancing 6G communications in the physical layer. *IEEE Wireless Commun.* **2021**, *28*, 141–147. [CrossRef]
- Song, H.J.; Lee, N. Terahertz Communications: Challenges in the Next Decade. *IEEE Trans. Terahertz Sci. Technol.* 2022, 12, 105–117. [CrossRef]
- Yang, N.; Shafie, A. Terahertz Communications for Massive Connectivity and Security in 6G and Beyond Era. *IEEE Commun. Mag.* 2022, 1–7. [CrossRef]
- Wang, J.; Wang, C.X.; Huang, J.; Chen, Y. 6G THz Propagation Channel Characteristics and Modeling: Recent Developments and Future Challenges. *IEEE Commun. Mag.* 2022, 1–8. [CrossRef]
- Akyildiz, I.F.; Han, C.; Hu, Z.; Nie, S.; Jornet, J.M. Terahertz band communication: An old problem revisited and research directions for the next decade. *IEEE Trans. Commun.* 2022, 70, 4250–4285. [CrossRef]
- Azari, M.M.; Solanki, S.; Chatzinotas, S.; Bennis, M. THz-Empowered UAVs in 6G: Opportunities, Challenges, and Trade-offs. IEEE Commun. Mag. 2022, 60, 24–30. [CrossRef]
- 32. Han, C.; Wang, Y.; Li, Y.; Chen, Y.; Abbasi, N.A.; Kürner, T.; Molisch, A.F. Terahertz Wireless Channels: A Holistic Survey on Measurement, Modeling, and Analysis. *IEEE Commun. Surve. Tutor.* **2022**, *24*, 1670–1707. [CrossRef]
- 33. Shafie, A.; Yang, N.; Han, C.; Jornet, J.M.; Juntti, M.; Kurner, T. Terahertz Communications for 6G and Beyond Wireless Networks: Challenges, Key Advancements, and Opportunities. *IEEE Netw.* **2022**, *70*, 1–8. [CrossRef]
- Khan, M.M.; Hossain, S.; Majumder, P.; Akter, S.; Ashique, R.H. A review on machine learning and deep learning for various antenna design applications. *Heliyon* 2022, 8, e09317. [CrossRef]
- Lu, X.; Xiao, L.; Li, P.; Ji, X.; Xu, C.; Yu, S.; Zhuang, W. Reinforcement Learning Based Physical Cross-Layer Security and Privacy in 6G. *IEEE Commun. Surv. Tutor.* 2022, 25, 425–466. [CrossRef]
- Helal, S.; Sarieddeen, H.; Dahrouj, H.; Al-Naffouri, T.Y.; Alouini, M.S. Signal Processing and Machine Learning Techniques for Terahertz Sensing: An overview. *IEEE Signal Proc. Mag.* 2022, 39, 42–62. [CrossRef]
- Muscinelli, E.; Shinde, S.S.; Tarchi, D. Overview of distributed machine learning techniques for 6G networks. *Algorithms* 2022, 15, 210. [CrossRef]
- Qurratulain Khan, M.; Gaber, A.; Schulz, P.; Fettweis, G. Machine Learning for Millimeter Wave and Terahertz Beam Management: A Survey and Open Challenges. *IEEE Access* 2023, 11, 11880–11902. [CrossRef]
- Gezimati, M.; Singh, G. Advances in terahertz technology for cancer detection applications. *Opt. Quantum Electron.* 2023, 55, 151. [CrossRef]
- Mchangama, A.; Ayadi, J.; Jiménez, V.P.G.; Consoli, A. MmWave massive MIMO small cells for 5G and beyond mobile networks: An overview. In Proceedings of the 12th International Symposium on Communication Systems, Networks and Digital Signal Processing (CSNDSP), Porto, Portugal, 20–22 July 2020; pp. 1–6. [CrossRef]
- Salahdine, F.; Han, T.; Zhang, N. 5G, 6G, and Beyond: Recent advances and future challenges. *Ann. Telecommun.* 2023, 1–25. [CrossRef]
- 42. Wang, X.; Kong, L.; Kong, F.; Qiu, F.; Xia, M.; Arnon, S.; Chen, G. Millimeter wave communication: A comprehensive survey. *IEEE Commun. Surv. Tutor.* **2018**, *20*, 1616–1653. [CrossRef]
- 43. Ullah, A.; Haq Abbas, Z.; Muhammad, F.; Abbas, G.; Jiao, L. Capacity driven small cell deployment in heterogeneous cellular networks: Outage probability and rate coverage analysis. *Trans. Emerg. Telecommun. Technol.* **2020**, *31*, e3876. [CrossRef]
- 44. Ullah, A.; Abbas, Z.H.; Muhammad, F.; Abbas, G.; Kim, S. Uplink performance analysis of user-centric small cell aided dense HCNets with uplink-downlink decoupling. *IEEE Access* 2020, *8*, 148460–148474. [CrossRef]
- Rappaport, T.S.; Xing, Y.; Kanhere, O.; Ju, S.; Madanayake, A.; Mandal, S.; Alkhateeb, A.; Trichopoulos, G.C. Wireless communications and applications above 100 GHz: Opportunities and challenges for 6G and beyond. *IEEE Access* 2019, 7, 78729– 78757. [CrossRef]
- Nie, S.; Jornet, J.M.; Akyildiz, I.F. Intelligent Environments Based on Ultra-massive Mimo Platforms for Wireless Communication in Millimeter Wave and Terahertz Bands. In Proceedings of the ICASSP 2019—2019 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), Brighton, UK, 12–17 May 2019; pp. 7849–7853. [CrossRef]
- Akyildiz, I.F.; Jornet, J.M.; Han, C. TeraNets: Ultra-broadband communication networks in the terahertz band. *IEEE Wirel. Commun.* 2014, 21, 130–135. [CrossRef]
- Sarkar, P.; Saha, A. Evolution of 6G and Terahertz Communication. In Optical to Terahertz Engineering; Springer: Berlin/Heidelberg, Germany, 2023; pp. 45–58.
- 49. Shen, X.; Gao, J.; Li, M.; Zhou, C.; Hu, S.; He, M.; Zhuang, W. Toward Immersive Communications in 6G. arXiv 2023, arXiv:2303.05283.
- IEEE Std 802.15.3d-2017; IEEE Standard for High Data Rate Wireless Multi-Media Networks–Amendment 2: 100 Gb/s Wireless Switched Point-to-Point Physical Layer. IEEE: Piscataway, NJ, USA, 2017; 1–55. [CrossRef]

- 51. Shehata, M.; Wang, K.; Webber, J.; Fujita, M.; Nagatsuma, T.; Withayachumnankul, W. IEEE 802.15.3d-Compliant Waveforms for Terahertz Wireless Communications. *J. Light. Technol.* **2021**, *39*, 7748–7760. [CrossRef]
- 52. Petrov, V.; Kurner, T.; Hosako, I. IEEE 802.15.3d: First Standardization Efforts for Sub-Terahertz Band Communications toward 6G. *IEEE Commun. Mag.* 2020, *58*, 28–33. [CrossRef]
- 53. Lu, X.; Venkatesh, S.; Saeidi, H. A review on applications of integrated terahertz systems. *China Commun.* **2021**, *18*, 175–201. [CrossRef]
- 54. Farhad, A.; Pyun, J.Y. Resource Management for Massive Internet of Things in IEEE 802.11 ah WLAN: Potentials, Current Solutions, and Open Challenges. *Sensors* 2022, 22, 9509. [CrossRef]
- Chen, H.; Sarieddeen, H.; Ballal, T.; Wymeersch, H.; Alouini, M.S.; Al-Naffouri, T.Y. A tutorial on terahertz-band localization for 6G communication systems. *IEEE Commun. Surv. Tutor.* 2022, 24, 1780–1815. [CrossRef]
- Akyildiz, I.F.; Gutierrez-Estevez, D.M.; Balakrishnan, R.; Chavarria-Reyes, E. LTE-Advanced and the evolution to Beyond 4G (B4G) systems. *Phys. Commun.* 2014, 10, 31–60. [CrossRef]
- 57. Akyildiz, I.F.; Jornet, J.M.; Han, C. Terahertz band: Next frontier for wireless communications. *Phys. Commun.* **2014**, *12*, 16–32. [CrossRef]
- IEEE Std 1906.1-2015; IEEE Recommended Practice for Nanoscale and Molecular Communication Framework. IEEE: Piscataway, NJ, USA, 2016; pp. 1–64. [CrossRef]
- 59. Kürner, T.; Priebe, S. Towards THz communications-status in research, standardization and regulation. *J. Infrared Millim. Terahertz Waves* **2014**, *35*, 53–62. [CrossRef]
- Yang, K.; Pellegrini, A.; Brizzi, A.; Alomainy, A.; Hao, Y. Numerical analysis of the communication channel path loss at the THz band inside the fat tissue. In Proceedings of the 2013 IEEE MTT-S International Microwave Workshop Series on RF and Wireless Technologies for Biomedical and Healthcare Applications (IMWS-BIO), Singapore, 9–11 December 2013; pp. 1–3. [CrossRef]
- Yang, K.; Alomainy, A.; Hao, Y. In-vivo characterisation and numerical analysis of the THz radio channel for nanoscale body-centric wireless networks. In Proceedings of the 2013 USNC-URSI Radio Science Meeting (Joint with AP-S Symposium), Lake Buena Vista, FL, USA, 7–13 July 2013; pp. 218–219. [CrossRef]
- IEEE Std 1906.1.1-2020; IEEE Standard Data Model for Nanoscale Communication Systems. IEEE: Piscataway, NJ, USA, 2020; pp. 1–142. [CrossRef]
- Canovas-Carrasco, S.; Garcia-Sanchez, A.J.; Garcia-Haro, J. The IEEE 1906.1 Standard: Nanocommunications as a new source of data. In Proceedings of the ITU Kaleidoscope: Challenges for a Data-Driven Society (ITU K), Nanjing, China, 27–29 November 2017; pp. 1–7. [CrossRef]
- 64. Yuan, Y.; Zhao, Y.; Zong, B.; Parolari, S. Potential key technologies for 6G mobile communications. *Sci. China Inf. Sci.* 2020, 63, 1–19. [CrossRef]
- Teng, Y.; Li, C.; Li, S.; Xiao, Y.; Jiang, L. Efficient design method for terahertz broadband metasurface patterns via deep learning. Optics Laser Technol. 2023, 160, 109058. [CrossRef]
- 66. Han, C.; Zhang, X.; Wang, X. On medium access control schemes for wireless networks in the millimeter-wave and terahertz bands. *Nano Commun. Netw.* **2019**, *19*, 67–80. [CrossRef]
- 67. Cacciapuoti, A.S.; Sankhe, K.; Caleffi, M.; Chowdhury, K.R. Beyond 5G: THz-based medium access protocol for mobile heterogeneous networks. *IEEE Commun. Mag.* 2018, *56*, 110–115. [CrossRef]
- 68. Polese, M.; Restuccia, F.; Gosain, A.; Jornet, J.; Bhardwaj, S.; Ariyarathna, V.; Mandal, S.; Zheng, K.; Dhananjay, A.; Mezzavilla, M.; et al. MillimeTera: Toward a large-scale open-source mmWave and terahertz experimental testbed. In Proceedings of the 3rd ACM Workshop on Millimeter-Wave Networks and Sensing Systems, Los Cabos, Mexico, 25 October 2019; pp. 27–32.
- Chen, H.; Zhang, G.; Li, X.; Zhu, P. Multi-agent Deep Reinforcement Learning Based Resource Management in IRS-NOMA Terahertz Network. In Proceedings of the 2022 IEEE/CIC International Conference on Communications in China (ICCC), Foshan, China, 11–13 August 2022; pp. 766–771. [CrossRef]
- Marnat, L.; Dussopt, L.; Puyal, V.; Siligaris, A.; Hameau, F.; Larie, A.; Dehos, C. V-band transceiver modules with integrated antennas and phased arrays for mmWave access in 5G mobile networks. In Proceedings of the 2017 11th European Conference on Antennas and Propagation (EUCAP), Paris, France, 19–24 March 2017; pp. 2786–2790.
- 71. Kutz, J.; Liebermeister, L.; Vieweg, N.; Wenzel, K.; Kohlhaas, R.; Naftaly, M. A Terahertz Fast-Sweep Optoelectronic Frequency-Domain Spectrometer: Calibration, Performance Tests, and Comparison with TDS and FDS. *Appl. Sci.* **2022**, *12*, 8257. [CrossRef]
- 72. Coutaz, J.L.; Garet, F.; Wallace, V. Principles of Terahertz Time-Domain Spectroscopy; CRC Press: Boca Raton, FL, USA, 2018.
- 73. Neu, J.; Schmuttenmaer, C.A. Tutorial: An introduction to terahertz time domain spectroscopy (THz-TDS). *J. Appl. Phys.* 2018, 124, 231101. [CrossRef]
- Withayachumnankul, W.; Naftaly, M. Fundamentals of measurement in terahertz time-domain spectroscopy. J. Infrared Millim. Terahertz Waves 2014, 35, 610–637. [CrossRef]
- Roh, W.; Seol, J.Y.; Park, J.; Lee, B.; Lee, J.; Kim, Y.; Cho, J.; Cheun, K.; Aryanfar, F. Millimeter-wave beamforming as an enabling technology for 5G cellular communications: Theoretical feasibility and prototype results. *IEEE Commun. Mag.* 2014, 52, 106–113. [CrossRef]
- 76. Completing the First Phase of the 5G Evolutions. Available online: https://www.qualcomm.com/content/dam/qcommmartech/dm-assets/documents/powerpoint_messaging_-_3gpp_release_17_completing_the_first_phase_of_5g_evolution. pdf (accessed on 8 May 2023).

- Kemmochi, F.; Fujisawa, K.; Otsuka, H. Potential Design for Modulation and Coding Scheme in mmWave Multicarrier HetNets. In Proceedings of the IEEE 90th Vehicular Technology Conference (VTC2019-Fall), Honolulu, HI, USA, 22–25 September 2019; pp. 1–5. [CrossRef]
- 78. El Jbari, M.; Moussaoui, M.; Chahboun, N. Coding and modulation for terahertz. In *Terahertz Wireless Communication Components and System Technologies*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 59–79.
- Zhang, H.; Venkateswaran, S.; Madhow, U. Channel modeling and MIMO capacity for outdoor millimeter wave links. In Proceedings of the 2010 IEEE Wireless Communication and Networking Conference, Sydney, Australia, 18–21 April 2010; pp. 1–6.
- Jalili, H.; Momeni, O. Scalable wideband and wide-angle beam steering mm-Wave/THz radiator and phased arrays in silicon. In Proceedings of the 2018 Asia-Pacific Microwave Conference (APMC), Kyoto, Japan, 6–9 November 2018; pp. 1157–1159.
- 81. Hemadeh, I.A.; Satyanarayana, K.; El-Hajjar, M.; Hanzo, L. Millimeter-wave communications: Physical channel models, design considerations, antenna constructions, and link-budget. *IEEE Commun. Surv. Tutor.* 2017, 20, 870–913. [CrossRef]
- 82. 5G; Study on Channel Model for Frequencies from 0.5 to 100 GHz (3GPP TR 38.901 Version 16.1.0 Release 16). Available online: https://www.etsi.org/deliver/etsi_tr/138900_138999/138901/16.01.00_60/tr_138901v160100p.pdf (accessed on 8 May 2023).
- Ju, S.; Xing, Y.; Kanhere, O.; Rappaport, T.S. Millimeter Wave and Sub-Terahertz Spatial Statistical Channel Model for an Indoor Office Building. *IEEE J. Sel. Areas Commun.* 2021, 39, 1561–1575. [CrossRef]
- Haneda, K.; Zhang, J.; Tan, L.; Liu, G.; Zheng, Y.; Asplund, H.; Li, J.; Wang, Y.; Steer, D.; Li, C.; et al. 5G 3GPP-Like Channel Models for Outdoor Urban Microcellular and Macrocellular Environments. In Proceedings of the IEEE 83rd Vehicular Technology Conference (VTC Spring), Nanjing, China, 15–18 May 2016; pp. 1–7. [CrossRef]
- Samimi, M.K.; Rappaport, T.S. 3-D Millimeter-Wave Statistical Channel Model for 5G Wireless System Design. *IEEE Transa.* Microw. Theory Tech. 2016, 64, 2207–2225. [CrossRef]
- 86. Rappaport, T.S.; MacCartney, G.R.; Samimi, M.K.; Sun, S. Wideband Millimeter-Wave Propagation Measurements and Channel Models for Future Wireless Communication System Design. *IEEE Trans. Commun.* **2015**, *63*, 3029–3056. [CrossRef]
- Maccartney, G.R.; Rappaport, T.S.; Sun, S.; Deng, S. Indoor Office Wideband Millimeter-Wave Propagation Measurements and Channel Models at 28 and 73 GHz for Ultra-Dense 5G Wireless Networks. *IEEE Access* 2015, *3*, 2388–2424. [CrossRef]
- Papasotiriou, E.N.; Boulogeorgos, A.A.A.; Haneda, K.; de Guzman, M.F.; Alexiou, A. An experimentally validated fading model for THz wireless systems. *Sci. Rep.* 2021, *11*, 18717. [CrossRef]
- Gomez-Ponce, J.; Abbasi, N.A.; Willner, A.E.; Zhang, C.J.; Molisch, A.F. Directionally Resolved Measurement and Modeling of THz Band Propagation Channels. *IEEE Open J. Antennas Propag.* 2022, 3, 663–686. [CrossRef]
- Gomez-Ponce, J.; Abbasi, N.A.; Cheng, Z.; Molisch, A.F. Directional characteristics of THz outdoor channels measurement and system performance implications. In Proceedings of the 2021 55th Asilomar Conference on Signals, Systems, and Computers, Pacific Grove, CA, USA, 31 October–3 November 2021; pp. 658–663. [CrossRef]
- 91. Gomez-Ponce, J.; Abbasi, N.A.; Kondaveti, R.; Kumar, A.; Abu-Surra, S.; Xu, G.; Zhang, C.; Molisch, A.F. Impact of common reflecting and absorbing building materials on THz multipath channels. *Radio Sci.* **2022**, *57*, 1–16. [CrossRef]
- Abbasi, N.A.; Gomez, J.L.; Kondaveti, R.; Shaikbepari, S.M.; Rao, S.; Abu-Surra, S.; Xu, G.; Zhang, J.; Molisch, A.F. THz Band Channel Measurements and Statistical Modeling for Urban D2D Environments. *IEEE Trans. Wireless Commun.* 2023, 22, 1466–1479. [CrossRef]
- Abbasi, N.A.; Hariharan, A.; Nair, A.M.; Almaiman, A.S.; Rottenberg, F.B.; Willner, A.E.; Molisch, A.F. Double Directional Channel Measurements for THz Communications in an Urban Environment. In Proceedings of the ICC 2020—2020 IEEE International Conference on Communications (ICC), Dublin, Ireland, 7–11 June 2020; pp. 1–6. [CrossRef]
- 94. 5G New Radio (NR): User Equipment (UE) Radio Transmission and Reception; Part 1: Range 1 Standalone (3GPP TS 38.101-1 version 15.2.0 Release 15). Available online: https://www.etsi.org/deliver/etsi_ts/138100_138199/13810101/15.02.00_60/ts_13 810101v150200p.pdf (accessed on 8 May 2023).
- 95. IEEE Std 802.11ad-2012 (Amendment to IEEE Std 802.11-2012, as amended by IEEE Std 802.11ae-2012 and IEEE Std 802.11aa-2012); IEEE Standard for Information technology–Telecommunications and information exchange between systems–Local and metropolitan area networks–Specific requirements-Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications Amendment 3: Enhancements for Very High Throughput in the 60 GHz Band. IEEE: Piscataway, NJ, USA, 2012; pp. 1–628. [CrossRef]
- 96. IEEE Std 802.11ay-2021 (Amendment to IEEE Std 802.11-2020 as amendment by IEEE Std 802.11ax-2021); IEEE Standard for Information Technology–Telecommunications and Information Exchange between Systems Local and Metropolitan Area Networks–Specific Requirements Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications Amendment 2: Enhanced Throughput for Operation in License-exempt Bands above 45 GHz. IEEE: Piscataway, NJ, USA, 2021; pp. 1–768. [CrossRef]
- Rupasinghe, N.; Kakishima, Y.; Guvenc, I.; Kitao, K.; Imai, T. Geometry performance for 5G mmWave cellular networks. In Proceedings of the 2016 International Symposium on Antennas and Propagation (ISAP), Okinawa, Japan, 24–28 October 2016; pp. 874–875.
- Talbi, D.; Gal, Z. Impact of Multi-Layer Recurrent Neural Networks in the Congestion Analysis of TeraHertz B5G/6G MAC Mechanism. In Proceedings of the 2022 International Conference on Software, Telecommunications and Computer Networks (SoftCOM), Split, Croatia, 22–24 September 2022; pp. 1–6. [CrossRef]

- He, L.; Hu, F.; Chu, Z.; Zhao, J.; Abuzaninab, N.; Sagduyu, Y.; Thawdar, N.; Kumar, S. Intelligent Terahertz Medium Access Control (MAC) for Highly Dynamic Airborne Networks. *IEEE Trans. Aerospace Elect. Syst.* 2022, 1–17. [CrossRef]
- Giordani, M.; Polese, M.; Roy, A.; Castor, D.; Zorzi, M. A tutorial on beam management for 3GPP NR at mmWave frequencies. IEEE Commun. Surv. Tutor. 2018, 21, 173–196. [CrossRef]
- Wang, H.; Zhang, P.; Li, J.; You, X. Radio propagation and wireless coverage of LSAA-based 5G millimeter-wave mobile communication systems. *China Commun.* 2019, 16, 1–18. [CrossRef]
- Cheng, M.; Wang, J.B.; Wu, Y.; Xia, X.G.; Wong, K.K.; Lin, M. Coverage Analysis for Millimeter Wave Cellular Networks With Imperfect Beam Alignment. *IEEE Trans. Vehicul. Technol.* 2018, 67, 8302–8314. [CrossRef]
- 103. Rappaport, T.S.; Sun, S.; Mayzus, R.; Zhao, H.; Azar, Y.; Wang, K.; Wong, G.N.; Schulz, J.K.; Samimi, M.; Gutierrez, F. Millimeter Wave Mobile Communications for 5G Cellular: It Will Work! *IEEE Access* 2013, 1, 335–349. [CrossRef]
- 104. Fujishima, M.; Amakawa, S.; Takano, K.; Katayama, K.; Yoshida, T. Tehrahertz CMOS design for low-power and high-speed wireless communication. *IEICE Trans. Electr.* **2015**, *98*, 1091–1104. [CrossRef]
- 105. Kallfass, I.; Dan, I.; Rey, S.; Harati, P.; Antes, J.; Tessmann, A.; Wagner, S.; Kuri, M.; Weber, R.; Massler, H.; et al. Towards MMIC-based 300GHz indoor wireless communication systems. *IEICE Trans. Electr.* 2015, 98, 1081–1090. [CrossRef]
- Mezzavilla, M.; Zhang, M.; Polese, M.; Ford, R.; Dutta, S.; Rangan, S.; Zorzi, M. End-to-end simulation of 5G mmWave networks. IEEE Commun. Surv. Tutor. 2018, 20, 2237–2263. [CrossRef]
- 107. Patriciello, N.; Lagen, S.; Bojovic, B.; Giupponi, L. An E2E simulator for 5G NR networks. *Simul. Model. Pract. Theory* **2019**, 96, 101933. [CrossRef]
- Tarboush, S.; Sarieddeen, H.; Chen, H.; Loukil, M.H.; Jemaa, H.; Alouini, M.S.; Al-Naffouri, T.Y. TeraMIMO: A channel simulator for wideband ultra-massive MIMO terahertz communications. *IEEE Trans. Veh. Technol.* 2021, 70, 12325–12341. [CrossRef]
- Ju, S.; Kanhere, O.; Xing, Y.; Rappaport, T.S. A Millimeter-Wave Channel Simulator NYUSIM with Spatial Consistency and Human Blockage. In Proceedings of the IEEE Global Communications Conference (GLOBECOM), Waikoloa, HI, USA, 9–13 December 2019; pp. 1–6. [CrossRef]
- 110. Yi, H.; Guan, K.; He, D.; Ai, B.; Dou, J.; Kim, J. Characterization for the vehicle-to-infrastructure channel in urban and highway scenarios at the terahertz band. *IEEE Access* 2019, 7, 166984–166996. [CrossRef]
- Piro, G.; Grieco, L.A.; Boggia, G.; Camarda, P. Nano-Sim: simulating electromagnetic-based nanonetworks in the network simulator 3. *SimuTools* 2013, 203–210. Available online: https://dl.acm.org/doi/abs/10.5555/2512734.2512762 (accessed on 25 April 2023).
- 112. Moshirfatemi, F. Communicating at Terahertz Frequencies. PhD Thesis, Portland State University, Portland, ON, USA, 2017.
- 113. Hossain, Z.; Xia, Q.; Jornet, J.M. TeraSim: An ns-3 extension to simulate terahertz-band communication networks. *Nano Commun. Netw.* **2018**, *17*, 36–44. [CrossRef]
- 114. Polese, M.; Jornet, J.M.; Melodia, T.; Zorzi, M. Toward end-to-end, full-stack 6G terahertz networks. *IEEE Commun. Mag.* 2020, 58, 48–54. [CrossRef]
- 115. Morales, D.; Jornet, J.M. Adapt: An adaptive directional antenna protocol for medium access control in terahertz communication networks. *Ad Hoc Netw.* **2021**, *119*, 102540. [CrossRef]
- Ma, X.; Chen, Z.; Chen, W.; Chi, Y.; Li, Z.; Han, C.; Wen, Q. Intelligent reflecting surface enhanced indoor terahertz communication systems. *Nano Commun. Netw.* 2020, 24, 100284. [CrossRef]
- 117. Davoudabadifarahani, H.; Ghalamkari, B. High efficiency miniaturized microstrip patch antenna for wideband terahertz communications applications. *Optik* **2019**, *194*, 163118. [CrossRef]
- 118. Ullah, A.; Abbas, Z.H.; Abbas, G.; Muhammad, F.; Kang, J.M. Hybrid millimeter wave heterogeneous networks with spatially correlated user equipment. *Digital Commun. Netw.* 2022, *in press.* [CrossRef]
- Simeone, O. A Very Brief Introduction to Machine Learning With Applications to Communication Systems. *IEEE Trans. Cogn. Commun. Netw.* 2018, 4, 648–664. [CrossRef]
- 120. Hussain, F.; Hassan, S.A.; Hussain, R.; Hossain, E. Machine Learning for Resource Management in Cellular and IoT Networks: Potentials, Current Solutions, and Open Challenges. *IEEE Commun. Surv. Tutor.* **2020**, *22*, 1251–1275. [CrossRef]
- 121. Hall, J.; Jornet, J.M.; Thawdar, N.; Melodia, T.; Restuccia, F. Deep Learning at the Physical Layer for Adaptive Terahertz Communications. *IEEE Trans. Terahertz Sci. Technol.* 2023, 13, 102–112. [CrossRef]
- Olson, N.; Andrews, J.G.; Heath, R.W. Coverage in Terahertz Cellular Networks with Imperfect Beam Alignment. In Proceedings of the IEEE Global Communications Conference (GLOBECOM), Madrid, Spain, 7–11 December 2021; pp. 1–6. [CrossRef]
- Moon, J.; Kim, S.; Ju, H.; Shim, B. Energy-Efficient User Association in mmWave/THz Ultra-Dense Network via Multi-Agent Deep Reinforcement Learning. *IEEE Trans. Green Commun. Netw.* 2023, 7, 692–706. [CrossRef]
- 124. Zarini, H.; Gholipoor, N.; Mili, M.R.; Rasti, M.; Tabassum, H.; Hossain, E. Resource Management for Multiplexing eMBB and URLLC Services Over RIS-Aided THz Communication. *IEEE Trans. Commun.* **2023**, 77, 1207–1225. [CrossRef]
- 125. Abuzainab, N.; Alrabeiah, M.; Alkhateeb, A.; Sagduyu, Y.E. Deep Learning for THz Drones with Flying Intelligent Surfaces: Beam and Handoff Prediction. In Proceedings of the 2021 IEEE International Conference on Communications Workshops (ICC Workshops), Montreal, QC, Canada, 14–23 June 2021; pp. 1–6. [CrossRef]
- 126. He, D.; Wang, Z. Deep Learning-Assisted Demodulation for TeraHertz Communications Under Hybrid Distortions. *IEEE Commun. Lett.* 2022, 26, 325–329. [CrossRef]

- 127. Shafie, A.; Lit, C.; Yang, N.; Zhou, X.; Duong, T.Q. An Unsupervised Learning Approach for Spectrum Allocation in Terahertz Communication Systems. In Proceedings of the GLOBECOM 2022—2022 IEEE Global Communications Conference, Rio de Janeiro, Brazil, 4–8 December 2022; pp. 3447–3453. [CrossRef]
- Hemant, K.; Bharadwaj, M.; Krishna, A.V. Automatic Modulation Classification for Terahertz Communication. In Proceedings of the 2022 International Conference on Wireless Communications Signal Processing and Networking (WiSPNET), Hyderabad, India, 19–22 December 2022; pp. 322–325. [CrossRef]
- Wu, Z.; Zhang, Y.; Li, D.; Ma, J. Modulation recognition of terahertz signals by deep learning. In Proceedings of the 2022 15th UK-Europe-China Workshop on Millimetre-Waves and Terahertz Technologies (UCMMT), Tonsberg, Norway, 17–18 October 2022; pp. 1–3. [CrossRef]
- Tong, X.; Chang, B.; Meng, Z.; Zhao, G.; Chen, Z. Calculating Terahertz Channel Capacity Under Beam Misalignment and User Mobility. *IEEE Wireless Commun. Lett.* 2022, 11, 348–351. [CrossRef]
- Moorthy, S.K.; Guan, Z. LeTera: Stochastic Beam Control Through ESN Learning in Terahertz-Band Wireless UAV Networks. In Proceedings of the IEEE INFOCOM 2020—IEEE Conference on Computer Communications Workshops (INFOCOM WKSHPS), Toronto, ON, Canada, 6–9 July 2020; pp. 1039–1044. [CrossRef]
- Huang, C.; Yang, Z.; Alexandropoulos, G.C.; Xiong, K.; Wei, L.; Yuen, C.; Zhang, Z. Hybrid Beamforming for RIS-Empowered Multi-hop Terahertz Communications: A DRL-based Method. In Proceedings of the 2020 IEEE Globecom Workshops (GC Wkshps, 2020), Virtual, 7–11 December 2020; pp. 1–6. [CrossRef]
- Barazideh, R.; Semiari, O.; Niknam, S.; Natarajan, B. Reinforcement Learning for Mitigating Intermittent Interference in Terahertz Communication Networks. In Proceedings of the 2020 IEEE International Conference on Communications Workshops (ICC Workshops), Dublin, Ireland, 7–11 June 2020; pp. 1–6. [CrossRef]
- Ahmadi, A.; Semiari, O. Reinforcement Learning for Optimized Beam Training in Multi-Hop Terahertz Communications. In Proceedings of the ICC 2021—IEEE International Conference on Communications, Virtual, 13–16 December 2021; pp. 1–6. [CrossRef]
- Hsu, P.C.; Shen, L.H.; Liu, C.H.; Feng, K.T. Federated Deep Reinforcement Learning for THz-Beam Search with Limited CSI. In Proceedings of the 2022 IEEE 96th Vehicular Technology Conference (VTC2022-Fall), London, UK, 26–29 September 2022; pp. 1–6. [CrossRef]
- Guo, Q.; Tang, F.; Kato, N. Federated Reinforcement Learning-Based Resource Allocation in D2D-Enabled 6G. IEEE Netw. 2022, 1–7. [CrossRef]
- 137. Wang, H.; Zhang, H.; Liu, X.; Long, K.; Nallanathan, A. Joint UAV Placement Optimization, Resource Allocation, and Computation Offloading for THz Band: A DRL Approach. *IEEE Trans. Wireless Commun.* **2022** . [CrossRef]
- 138. Garcia-Sanchez, A.J.; Asorey-Cacheda, R.; Garcia-Haro, J.; Gomez-Tornero, J.L. Dynamic Multi-Hop Routing in Terahertz Flow-guided Nanosensor Networks: A Reinforcement Learning Approach. *IEEE Sens. J.* **2023**, *23*, 3408–3422. [CrossRef]
- Koenig, S.; Lopez-Diaz, D.; Antes, J.; Boes, F.; Henneberger, R.; Leuther, A.; Tessmann, A.; Schmogrow, R.; Hillerkuss, D.; Palmer, R.; et al. Wireless sub-THz communication system with high data rate. *Nat. Photonics* 2013, 7, 977–981. [CrossRef]
- Kallfass, I.; Boes, F.; Messinger, T.; Antes, J.; Inam, A.; Lewark, U.; Tessmann, A.; Henneberger, R. 64 Gbit/s transmission over 850 m fixed wireless link at 240 GHz carrier frequency. J. Infrared Millim. Terahertz Waves 2015, 36, 221–233. [CrossRef]
- 141. Yu, X.; Jia, S.; Hu, H.; Galili, M.; Morioka, T.; Jepsen, P.U.; Oxenløwe, L.K. 160 Gbit/s photonics wireless transmission in the 300-500 GHz band. *APL Photonics* **2016**, *1*, 081301. [CrossRef]
- 142. Deal, W.R.; Foster, T.; Wong, M.B.; Dion, M.; Leong, K.; Mei, X.B.; Zamora, A.; Altvater, G.; Kanemori, K.; Christen, L.; et al. A 666 GHz demonstration crosslink with 9.5 Gbps data rate. In Proceedings of the IEEE MTT-S International Microwave Symposium (IMS), Honololu, HI, USA, 4–9 June 2017; pp. 233–235. [CrossRef]
- 143. Sen, P.; Pados, D.A.; Batalama, S.N.; Einarsson, E.; Bird, J.P.; Jornet, J.M. The TeraNova platform: An integrated testbed for ultra-broadband wireless communications at true Terahertz frequencies. *Comp. Netw.* **2020**, *179*, 107370. [CrossRef]
- 144. Sen, P.; Ariyarathna, V.; Madanayake, A.; Jornet, J.M. A versatile experimental testbed for ultrabroadband communication networks above 100 GHz. *Comp. Netw.* 2021, 193, 108092. [CrossRef]
- 145. Gezimati, M.; Singh, G. Terahertz Imaging and Sensing for Healthcare: Current Status and Future Perspectives. *IEEE Access* 2023, *11*, 18590–18619. [CrossRef]
- 146. Brückner, C.; Pradarutti, B.; Müller, R.; Riehemann, S.; Notni, G.; Tünnermann, A. Design and evaluation of a THz time domain imaging system using standard optical design software. *Appl. Optics* **2008**, *47*, 4994–5006. [CrossRef]
- 147. Corsi, C.; Sizov, F. *THz and Security Applications: Detectors, Sources and Associated Electronics for THz Spplications;* Springer: Berlin/Heidelberg, Germany, 2014.
- 148. Li, M.; Liu, Z.; Xia, Y.; He, M.; Yang, K.; Yuan, S.; Yan, M.; Huang, K.; Zeng, H. Terahertz Time-of-Flight Ranging with Adaptive Clock Asynchronous Optical Sampling. *Sensors* 2023, 23, 715. [CrossRef]
- 149. Terahertz Wave Spectroscopy and Imaging Analysis Platform. Available online: https://https://www.advantest.com/products/ terahertz-imaging/platform.html (accessed on 25 April 2023).
- 150. Yang, F.; Pitchappa, P.; Wang, N. Terahertz reconfigurable intelligent surfaces (RISs) for 6G communication links. *Micromachines* **2022**, *13*, 285. [CrossRef]

- Rappaport, T.S.; Sun, S.; Shafi, M. Investigation and Comparison of 3GPP and NYUSIM Channel Models for 5G Wireless Communications. In Proceedings of the IEEE 86th Vehicular Technology Conference (VTC-Fall), Toronto, ON, Canada, 24–27 September 2017; pp. 1–5. [CrossRef]
- 152. NYUSIM 4.0, An Indoor Channel Simulator Up to 150 GHz. Available online: https://wireless.engineering.nyu.edu/nyusim/ (accessed on 6 May 2023).
- Sun, S.; MacCartney, G.R.; Rappaport, T.S. A novel millimeter-wave channel simulator and applications for 5G wireless communications. In Proceedings of the 2017 IEEE International Conference on Communications (ICC), Paris, France, 21–25 May 2017; pp. 1–7. [CrossRef]
- 154. He, D.; Ai, B.; Guan, K.; Wang, L.; Zhong, Z.; Kürner, T. The design and applications of high-performance ray-tracing simulation platform for 5G and beyond wireless communications: A tutorial. *IEEE Commun. Surv. Tutor.* **2018**, *21*, 10–27. [CrossRef]
- 155. Guan, K.; Peng, B.; He, D.; Eckhardt, J.M.; Rey, S.; Ai, B.; Zhong, Z.; Kürner, T. Measurement, simulation, and characterization of train-to-infrastructure inside-station channel at the terahertz band. *IEEE Trans. Terahertz Sci. Technol.* 2019, 9, 291–306. [CrossRef]
- 156. Priebe, S.; Kurner, T. Stochastic modeling of THz indoor radio channels. *IEEE Trans. Wirel. Commun.* **2013**, *12*, 4445–4455. [CrossRef]
- 157. Guan, K.; Zhong, Z.; Ai, B.; Kurner, T. Deterministic Propagation Modeling for the Realistic High-Speed Railway Environment. In Proceedings of the IEEE 77th Vehicular Technology Conference (VTC Spring), Dresden, Germany, 2–5 June 2013; pp. 1–5. [CrossRef]
- 158. He, D.; Guan, K.; Fricke, A.; Ai, B.; He, R.; Zhong, Z.; Kasamatsu, A.; Hosako, I.; Kürner, T. Stochastic channel modeling for kiosk applications in the terahertz band. *IEEE Trans. Terahertz Sci. Technol.* **2017**, *7*, 502–513. [CrossRef]
- 159. Moshir, F.; Singh, S. Pulsed terahertz time-domain communication. In Proceedings of the IEEE Global Communications Conference, Austin, TX, USA, 8–12 December 2014; pp. 3796–3801. [CrossRef]
- 160. TeraSim—An ns-3 Module for THz Networks. Available online: https://github.com/UN-Lab/thz, (accessed on 03 May, 2023).
- 161. Jornet, J.M.; Akyildiz, I.F. Channel Modeling and Capacity Analysis for Electromagnetic Wireless Nanonetworks in the Terahertz Band. *IEEE Trans. Wireless Commun.* 2011, 10, 3211–3221. [CrossRef]
- 162. Jornet, J.M.; Akyildiz, I.F. Femtosecond-long pulse-based modulation for terahertz band communication in nanonetworks. *IEEE Trans. Commun.* **2014**, *62*, 1742–1754. [CrossRef]
- Xia, Q.; Hossain, Z.; Medley, M.; Jornet, J.M. A Link-Layer Synchronization and Medium Access Control Protocol for Terahertz-Band Communication Networks. In Proceedings of the IEEE Global Communications Conference (GLOBECOM), San Diego, CA, USA, 6–10 December 2015; pp. 1–7. [CrossRef]
- 164. Mobility Models in ns-3. Available online: https://www.nsnam.org/docs/models/html/mobility.html (accessed on 28 December 2022).
- Farhad, A.; Kim, D.H.; Kim, B.H.; Mohammed, A.F.Y.; Pyun, J.Y. Mobility-Aware Resource Assignment to IoT Applications in Long-Range Wide Area Networks. *IEEE Access* 2020, *8*, 186111–186124. [CrossRef]
- Farhad, A.; Kim, D.H.; Pyun, J.Y. R-ARM: Retransmission-Assisted Resource Management in LoRaWAN for the Internet of Things. *IEEE Internet Things J.* 2022, 9, 7347–7361. [CrossRef]
- 167. Ning, B.; Tian, Z.; Mei, W.; Chen, Z.; Han, C.; Li, S.; Yuan, J.; Zhang, R. Beamforming Technologies for Ultra-Massive MIMO in Terahertz Communications. *IEEE Open J. Commun. Soc.* **2023**, *4*, 614–658. [CrossRef]
- 168. Farhad, A.; Kim, D.H.; Pyun, J.Y. Scalability of LoRaWAN in an Urban Environment: A Simulation Study. In Proceedings of the Eleventh International Conference on Ubiquitous and Future Networks (ICUFN), Zagreb, Croatia, 2–5 July 2019; pp. 677–681. [CrossRef]
- 169. Featured Building Parameters for Urban Environment in ns-3. Available online: https://www.nsnam.org/docs/release/3.19 /doxygen/classns3_1_1_building.html (accessed on 28 December 2022).
- 170. Boulogeorgos, A.A.A.; Yaqub, E.; Desai, R.; Sanguanpuak, T.; Katzouris, N.; Lazarakis, F.; Alexiou, A.; Di Renzo, M. Artificial Intelligence Empowered Multiple Access for Ultra Reliable and Low Latency THz Wireless Networks. *arXiv* 2022, arXiv:2208.08039.
- 171. Liu, Y.; Deng, Y.; Nallanathan, A.; Yuan, J. Machine Learning for 6G Enhanced Ultra-Reliable and Low-Latency Services. *IEEE Wireless Commun.* 2023, *30*, 48–54. [CrossRef]
- Farhad, A.; Kim, D.H.; Yoon, J.S.; Pyun, J.Y. Deep Learning-Based Channel Adaptive Resource Allocation in LoRaWAN. In Proceedings of the 2022 International Conference on Electronics, Information, and Communication (ICEIC), Jeju, Republic of Korea, 6–9 February 2022; pp. 1–5. [CrossRef]
- 173. Farhad, A.; Pyun, J.Y. AI-ERA: Artificial Intelligence-Empowered Resource Allocation for LoRa-Enabled IoT Applications. *IEEE Trans. Ind. Inf.* 2023, 1–13. [CrossRef]
- 174. Open Neural Network Exchange (ONNX). Available online: https://github.com/onnx/onnx (accessed on 28 December 2022).
- 175. Qian, L.; Yang, P.; Xiao, M.; Dobre, O.A.; Renzo, M.D.; Li, J.; Han, Z.; Yi, Q.; Zhao, J. Distributed Learning for Wireless Communications: Methods, Applications and Challenges. *IEEE J. Sel. Top. Signal Process.* **2022**, *16*, 326–342. [CrossRef]
- 176. Foukalas, F. Federated-Learning-Driven Radio Access Networks. IEEE Wireless Commun. 2022, 29, 48–54. [CrossRef]
- Liu, Y.J.; Qin, S.; Sun, Y.; Feng, G. Resource Consumption for Supporting Federated Learning in Wireless Networks. *IEEE Trans. Wireless Commun.* 2022, 21, 9974–9989. [CrossRef]

- 178. Xie, B.; Sun, Y.; Zhou, S.; Niu, Z.; Xu, Y.; Chen, J.; Gündüz, D. MOB-FL: Mobility-Aware Federated Learning for Intelligent Connected Vehicles. *arXiv* 2022, arXiv:2212.03519.
- 179. Yang, Z.; Chen, M.; Wong, K.K.; Poor, H.V.; Cui, S. Federated learning for 6G: Applications, challenges, and opportunities. *Engineering* **2022**, *8*, 33–41. [CrossRef]
- Letaief, K.B.; Shi, Y.; Lu, J.; Lu, J. Edge artificial intelligence for 6G: Vision, enabling technologies, and applications. *IEEE J. Sel. Areas Commun.* 2021, 40, 5–36. [CrossRef]
- 181. Patil, A.; Iyer, S.; Pandya, R.J. A survey of machine learning algorithms for 6g wireless networks. arXiv 2022, arXiv:2203.08429.
- 182. Xian, W.; Yu, K.; Han, F.; Fang, L.; He, D.; Han, Q.L. Advanced Manufacturing in Industry 5.0: A Survey of Key Enabling Technologies and Future Trends. *IEEE Trans. Ind. Inf.* **2023**, 1–15. [CrossRef]
- 183. Xue, Q.; Liu, Y.J.; Sun, Y.; Wang, J.; Yan, L.; Feng, G.; Ma, S. Beam Management in Ultra-Dense mmWave Network via Federated Reinforcement Learning: An Intelligent and Secure Approach. *IEEE Trans. Cogn. Commun. Netw.* **2023**, *9*, 185–197. [CrossRef]
- Ekaireb, E.; Yu, X.; Ergun, K.; Zhao, Q.; Lee, K.; Huzaifa, M.; Rosing, T. ns3-fl: Simulating Federated Learning with ns-3. In Proceedings of the 2022 Workshop on ns-3, Virtual, 22–23 June 2022; pp. 97–104.
- 3GPP. NS3-FL: Federated Learning Simulator. 2020. Available online: https://github.com/eekaireb/ns3-fl-network/tree/ec427 6cc861f2ff5afdba23571790b8783d56790 (accessed on 7 May 2023).
- Nisar, M.Z.; Ibrahim, M.S.; Usman, M.; Lee, J.A. A Lightweight Deep Learning Model for Automatic Modulation Classification Using Residual Learning and Squeeze–Excitation Blocks. *Appl. Sci.* 2023, 13, 5145. [CrossRef]
- 187. ns3-Gym (opengym) Module for Reinforcement Learning. Available online: https://github.com/tkn-tub/ns3-gym (accessed on 18 January 2023).
- Gawłowicz, P.; Zubow, A. ns-3 meets OpenAI Gym: The Playground for Machine Learning in Networking Research. In Proceedings of the ACM International Conference on Modeling, Analysis and Simulation of Wireless and Mobile Systems (MSWiM), Miami, FL, USA, 25–29 November 2019.
- 189. ns3-AI Module. Available online: https://github.com/hust-diangroup/ns3-ai (accessed on 18 January 2023).
- Yin, H.; Liu, P.; Liu, K.; Cao, L.; Zhang, L.; Gao, Y.; Hei, X. Ns3-Ai: Fostering Artificial Intelligence Algorithms for Networking Research. In Proceedings of the 2020 Workshop on Ns-3; Association for Computing Machinery, WNS3 2020, New York, NY, USA, 17–18 June 2020; pp. 57–64. [CrossRef]
- 191. API for Reinforcement Learning. Available online: https://www.gymlibrary.dev/i (accessed on 18 January 2023).
- 192. Sheikh, S.M.; Asif, H.M.; Raahemifar, K.; Al-Turjman, F. Time Difference of Arrival Based Indoor Positioning System Using Visible Light Communication. *IEEE Access* 2021, *9*, 52113–52124. [CrossRef]
- Subedi, S.; Hwang, S.S.; Pyun, J.Y. Hybrid Wireless Indoor Positioning System Combining BLE Beacons And Wi-Fi APs. In Proceedings of the International Conference on Information and Communication Technology Convergence (ICTC), Jeju, Republic of Korea, 21–23 October 2020; pp. 36–41. [CrossRef]
- 194. Kim, D.H.; Farhad, A.; Pyun, J.Y. UWB Positioning System Based on LSTM Classification With Mitigated NLOS Effects. IEEE Internet Things J. 2023, 10, 1822–1835. [CrossRef]
- Subedi, S.; Kim, D.H.; Kim, B.H.; Pyun, J.Y. Improved Smartphone-Based Indoor Localization System Using Lightweight Fingerprinting and Inertial Sensors. *IEEE Access* 2021, 9, 53343–53357. [CrossRef]
- Bhattarai, B.; Yadav, R.K.; Gang, H.S.; Pyun, J.Y. Geomagnetic Field Based Indoor Landmark Classification Using Deep Learning. IEEE Access 2019, 7, 33943–33956. [CrossRef]
- Fan, S.; Wu, Y.; Han, C.; Wang, X. A Structured Bidirectional LSTM Deep Learning Method For 3D Terahertz Indoor Localization. In Proceedings of the IEEE INFOCOM 2020—IEEE Conference on Computer Communications, Toronto, ON, Canada, 6–9 July 2020; pp. 2381–2390. [CrossRef]
- 198. Fan, S.; Wu, Y.; Han, C.; Wang, X. SIABR: A Structured Intra-Attention Bidirectional Recurrent Deep Learning Method for Ultra-Accurate Terahertz Indoor Localization. *IEEE J. Sel. Areas Commun.* **2021**, *39*, 2226–2240. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.